

# Automated Sleep Stage Detection and

# **Classification of Sleep Disorders**

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

by

Emad Abdulhadi Malaekah MSc, BSc

Electrical and Computer Engineering College of Science, Engineering and Health RMIT University

January 2016

### DECLARATION

I certify that except where due acknowledgement has been made, the work is that of the author alone; the work has not been submitted previously in whole or in part to qualify for any other academic award; the content of the thesis is the result of work that has been carried out since the official commencement date of the approved research program; any editorial work, paid or unpaid, carried out by a third party is acknowledged; and ethics procedures and guidelines have been followed.

Emad Abdulhadi Malaekah

06/Apr/2016

### ACKNOWLEDGMENT

I am grateful to my supervisor, Dr. Dean Cvetkovic, for his support and understanding throughout the research. I wish to express sincere appreciation to my co-supervisor, Associate Professor Margaret Lech, for her support. I would like to express my gratitude to Dr. Elena Pirogova for her cooperation and guidance. I gratefully acknowledge the financial support of the Ministry of Higher Education of the Kingdom of Saudi Arabia. I would also like to thank Stephanie Devuyst and the Sleep Laboratory of the Central Institute of Mental Health for providing the sleep PSG data, which served as the experimental source for the present research.

I would like to thank all of my friends in Saudi Arabia and Australia who continuously offered me help, encouragement and advice.

Finally, I am indebted to my wife, Rabab, my daughter, Zahra, and my son, Abdulhadi, for all the love and support given to me, and for their patience.

## DEDICATION

I dedicate this dissertation to my parents, brothers and sisters who taught me inquisitiveness and persistence, and to my wife and children who have stood with me during this long and challenging endeavour.

### ABSTRACT

Studies have demonstrated that more than 1 million Australians experience some sort of sleep-related disorder in their lifetime [12]. In order to improve the diagnostic and clinical treatment of sleep disorders, the first important step is to identify or automatically detect the sleep stages. The most common method, known as the visual sleep stage scoring, can be a tedious and time-consuming process. Because of that, there is a need to create or develop an improved automatic sleep stage detection method to assist the sleep physician to efficiently and accurately evaluate the sleep stages of patients or non-patients.

This research project consisted of two parts. The first part focused on the automatic sleep stages detection based on two individual bio-signals, which made up an overnight polysomnography (PSG), such as the electroencephalogram (EEG), and electrooculogram (EOG). Several features were extracted from these two bio-signals in the time and frequency domains. The decision tree and classification methods were utilised for the classification of the sleep stages.

The second part of this project focused on the automatic classification of different sleep and psychiatric disorders, such as patients with periodic limb movements of sleep (PLMs), sleep apnea-hypopnea syndrome (SAHS), primary insomnia, schizophrenia and healthy sleep. Different PSG parameters were computed for the classification of sleep disorders, such as descriptive statistics of sleep architecture. In conclusion, the advantage of an automatic sleep stage detection method based on a single-channel EEG or EOG signal can be undertaken with portable sleep stage recording instead of full the PSG system, which includes multichannel bio-signals. An automatic classification method of sleep and psychiatric disorders based on the descriptive statistics of sleep architecture statistics was found to be an effective technique for screening sleep and psychiatric disorders. This classification method can assist physicians to quickly undertake a diagnostic procedure.

# Table of Contents

DECLARATION	2
ACKNOWLEDGMENT	3
DEDICATION	4
ABSTRACT	5
LIST OF PUBLICATIONS	11
LIST OF TABLES	13
LIST OF FIGURES	15
LIST OF ABBREVIATIONS	18
1 : Introduction	22
1.1 Introduction	23
1.2 Motivation	25
1.3 Research aims	26
1.4 Objectives	27
1.5 Thesis composition	27
2 : Literature Review	29
2.1 Basis of Sleep	
2.1.1 Non-Rapid Eye Moment Sleep	31
2.1.2 Rapid Eye Movement Sleep	31
2.2 Sleep disorders and psychiatric disorder	32
2.2.1 Primary insomnia	
2.2.2 Periodic limb movement syndrome	35

2.2.3 Obstructive sleep apnea-hypopnea syndrome	35
2.2.4 Schizophrenia	37
2.2.5 Classification of sleep disorders	38
2.3 Visual sleep stage scoring	39
2.3.1 The Rechtscaffen and Kales scoring rules	40
2.3.2 The American Academy of Sleep Medicine scoring rules	41
2.4 Polysomnography and Hypnogram	41
2.4.1 Electroencephalography (EEG)	44
2.4.2 Electrooculogram	47
2.5 Automatic sleep stage detection	50
3 : Subjects and data sets	62
3.1 Subjects	63
3.1.1 Data set No.1	63
3.1.2 Data set No.2	64
3.1.3 Data set No.3	65
4 : Automatic sleep stage detection methods	67
4.1 Automatic detection of wake and sleep stage N1 using EEG sub-epoch	
approach	68
4.1.1 Subjects	68
4.1.2 EEG signal processing method	68
4.2 Automatic detection of sleep stages using EEG sub-epoch approach	71
4.2.1 Subjects	71

4.2.2 Automatic sleep stages detection algorithm	71
4.3 Automatic detection of sleep stages using EOG	78
4.3.1 Subjects	78
4.3.2 EOG signal Method	78
4.4 Sleep architecture statistical method	81
5 : Automatic classification of sleep disorders	83
5.1 Classification of primary insomnia, schizophrenia and healthy sleep	84
5.1.1 Subjects	84
5.1.2 Method	84
5.2 Classification of PLMS, SAHS and healthy sleep	85
5.2.1 Subjects	85
5.2.2 Method	86
6 : Results	88
6.1 Automatic detection of wake and sleep stage N1 using EEG sub-epoch	
approach	89
6.2 Automatic detection of sleep stages using EEG sub-epoch approach	92
6.2.1 Overall Performance	92
6.2.2 Comparison of sleep stage detection method	93
6.3 Individual subject performance	94
6.4 Automatic detection of sleep stages using EOG	97
6.5 Classification of insomnia, schizophrenia and healthy sleep	101
6.6 Classification of PLMS, SAHS and healthy sleep	103

7 : Discussion
7.1 Automatic detection of wake and sleep stage N1 using the EEG sub-epoch
approach107
7.2 Automatic detection of sleep stages using EEG sub-epoch approach
7.3 Automatic detection of sleep stages using EOG and classification of PLMS,
SAHS and healthy sleep110
7.4 Classification of insomnia, schizophrenia and healthy sleep113
8 : Conclusion and future work115
8.1 Conclusion
8.2 Future Work
REFERENCES

### LIST OF PUBLICATIONS

### Journal papers

- E. Malaekah, S. Shahrbabaki and D. Cvetkovic, "Automatic sleep stage detection and classification Periodic Limb Movements of Sleep, Sleep Apnea Hypopnea Syndrome and healthy control based on of electrooculography (EOG) signals", Journal of Bioprocessing and Biotechniques, 2015, 5:216.
- E. Malaekah, and D. Cvetkovic, "Automatic sleep stage detection with high sensitivity on primary insomnia and schizophrenia disorders" Journal of Artificial Intelligence in Medicine (under review).

### **Conference papers**

- E. Malaekah and D. Cvetkovic, "Automatic detection of the wake and stage 1 sleep stages using the EEG sub-epoch approach", in Engineering in Medicine and Biology Society (EMBC), 35th Annual International Conference of the IEEE, 2013, japaj,pp. 6401-6404.
- E. Malaekah and D. Cvetkovic, "Automatic sleep stage detection using consecutive and non-consecutive approach for elderly and young healthy subject", in Biosignals and Biorobotics Conference: Biosignals and Robotics for Better and Safer Living (BRC), 5th ISSNIP-IEEE, 2014, pp. 1-6.
- E. Malaekah, C. Patti and D. Cvetkovic," Automatic Sleep-Wake Detection using Electrooculogram Signals", in Biomedical Engineering and Sciences (IECBES), 2014 IEEE EMBS Conference, 2014.
- 6. E. Malaekah, H. Abdullh and D. Cvetkovic," Automatic Sleep stage Detection based on Electrooculogram" in Medical and Biological Engineering & Computing

(MBEC), 2015 (accepted).

- E. Malaekah, M. Schredl and D. Cvetkovic, "Automatic classification of insomnia, schizophrenia and healthy sleep based on polysomnographic sleep-stage", Australian Chronobiology Society (ACS) 11th Annual Meeting, Melbourne, Australia, November, 2014.
- R. Chaparro-Vargas, E. Malaekah, C. Schilling, M. Schredl, and D. Cvetkovic, "Interdependence of electroencephalographic and electrocardiographic power bands in human sleep", in Biosignals and Biorobotics Conference (2014): Biosignals and Robotics for Better and Safer Living (BRC), 5th ISSNIP-IEEE, 2014, pp. 1-6.

# LIST OF TABLES

Table 1: Polysomnographic features for primary insomnia disorders	.34
Table 2: Polysomnographic features of PLMs syndrome	.35
Table 3: Polysomnographic features of Obstructive sleep apnea	37
Table 4: Polysomnographic features of schizophrenia disorders	38
Table 5: The recommend parameters to be reported for PSG during sleep [13]	42
Table 6: NREM sleep phases and their characteristics [95].	45
Table 7: An overview of automatic sleep staging methods and studies based on	
feature extraction, classifier, number of subjects, accuracy and signals	60
Table 8: Summary of the PSG napping data	64
Table 9: Summary of PSG data set 2.	65
Table 10: Summary of PSG data set 3.	.66
Table 11: Sleep stage algorithm rules.	76
Table 12: The smoothing rules	78
Table 13: The count of W and S1 stages scored epochs from the three automated	
detection methods using parametric and non-parametric spectral density functions,	
and compared with the commercial 'hypnolab' sleep scoring	.90
Table 14: The comparison of the accuracy between the three automated detection	
methods using non-parametric and parametric spectral density functions	92
Table 15: The overall performance and comparison between visual and automatic	
detection (by both 3 consecutive and non-consecutive sub-epoch approaches) for th	ıe
three groups	94
Table 16: The confusion matrix of a single healthy (control) subject.	.98
Table 17: The confusion matrix of a single PLMS patient	98
Table 18: The confusion matrix of a single SAHS patient.	98

Table 19: The post-hoc t-tests of the differences in the sleep parameters between the
three groups102
Table 20: Confusion matrix of detection of the three participant groups based on the
decision three analysis
Table 21: The post-hoc t-tests of the differences in the sleep parameters between the
three patient groups104
Table 22: Confusion matrix of detection of the PLM and SAHS patients and healthy
subjects based on the decision three analysis
Table 23: A comparison of candidature own approach with existing commercial
algorithms based on the single EEG channel109

# LIST OF FIGURES

Figure 1: Hypnogram sample, where the y-axis indicates the sleep stages and the x-
axis indicates time (epoch) [95]44
Figure 2: EEG frequency bands, where the y-axis indicates EEG frequency bands and
the x-axis indicates time per second (sec) of the brain [101]45
Figure 3: The EEG electrode positions of the 10-20 system [101]47
Figure 4: Recommended, previous and alternate electrode locations of the EOG
signal. LOC = left outer canthus; ROC = right outer canthus [95]48
Figure 5: The automatic analysis process [130]
Figure 6: The hypnogram scoring of different human experts vs the Somnolyzer $24 \times 7$
system. The first hypnogram is indicate to the first expert scorer of sleep stage, the
second hypnogarm is the second expert scorer of sleep stage, the third hypnogram is
indicate to the consensus scorer and the final hypnogram is indicate to the
Somnolyzer 24×7 system scoring [150]
Figure 7: Recording from group (a) percentages of slow wave (SW2 where 2 refers to
rule number 2) segments; (b) SW3 (using rule 3) percentages; (c) percentages of rule
S; (d) percentages of rule S; (e) automatic sleep stage scoring developed; (f) visual
sleep scoring
Figure 8: Flow-chart describing the consecutive and non-consecutive, 6-second EEG
epoch approach
Figure 9: Flow chart describing the automated sleep stage detection algorithm based
on the three consecutive and non-consecutive 6-second EEG sub-epochs approach.
Where the '*' is indicated to 3 consecutive or non-consecutive out 5 sub-epoch
approach72

Figure 10: Flow chart depicting process for sleep stage identification applied for both Figure 11: Block diagram of the automatic sleep stage detection method using EOG Figure 12: Flow chart describing the automatic classification algorithm for primary insomnia, schizophrenia and healthy (control) patients based on the decision tree analysis. R1 is characterised by the first REM period in minutes, R2 by the % of sleep Figure 13: Flow chart based on decision tree analysis describing the automatic classification algorithm for PLMS and SAHS patients, and healthy subjects. N1 and Figure 14: The hypnograms for subject S22 using the parametric approach, comparing the 30-second and the three consecutive and non-consecutive 6-second epochs Figure 15: Hypnogram of visual sleep stage scoring versus automatic sleep stage scoring using 30-second epoch, three consecutive 6-second sub-epoch and three non-Figure 16: Cohen's Kappa (C.K.) and accuracy (ACC) agreement of automatic detection approaches using 30-secend epoch, three consecutive 6-second and three non-consecutive 6-second methods with visual detection by expert for each primary Figure 17: Cohen's Kappa (C.K.) and Accuracy (ACC) agreement of automatic detection approaches using 30-second epoch, three consecutive 6-second and three non-consecutive 6-second methods with visual detection by expert for each schizophrenia patient (n = 10 subjects, Sub 1-10)......96

Figure 18: Cohen's Kappa (C.K.) and Accuracy (ACC) agreement of automatic
detection approaches using 30-second epoch, three consecutive 6-second and three
non-consecutive 6-second methods with visual detection by expert for each healthy
subject (n = 10 subjects, Sub 1-10)96
Figure 19: The hypnogram of visual sleep stage scoring vs. automatic scoring for a
single healthy (control) subject
Figure 20: The hypnogram of visual sleep stage scoring vs. automatic scoring for a
single PLMS patient
Figure 21: The hypnogram of visual sleep stage scoring vs. automatic scoring for an
SAHS patient
Figure 22: The accuracy of the selected features for the detection of sleep stages for
healthy subjects100
Figure 23: The accuracy of the selected features for detection sleep stages for PLMS
patients100
Figure 24: The accuracy of the selected features for the detection of sleep stages for
SAHS patients101
Figure 25: Bar plots of the different sleep parameters for the primary insomnia
patients, schizophrenia patients and the healthy subject. SWD = slow-wave sleep
duration; FPR = the first period REM; RED = the REM duration103
Figure 26: Bar plots of the different sleep parameters for the PLMS and SAHS
patients and healthy subjects

### LIST OF ABBREVIATIONS

- AASM American Academic Sleep Medicine
- AR Autoregressive
- ASDA American Sleep Disorders Association
- BP Back propagation
- C.K. Cohen's Kappa
- CF Central frequency
- CIC Cascaded integrator-Comb
- DDTW Derivative dynamic time warping
- DREM Density of rapid eye movement
- DSMMD Diagnostic and Statistical Manual of Mental Disorders
- EDF European data format
- EDS Excessive daytime sleepiness
- EMD Empirical mode decomposition
- FFT Fast fourier transform
- FRP First REM period
- HHT Hilbert-Huang transform
- HMM Hidden Markov models

- ICD International Classification of Diseases
- ICSD International Classification of Sleep Disorders
- IRA Inter-rate agreement
- KC K complexes
- KNN K-nearest neighbour
- LMS Least mean square
- LOC Left outer canthus
- MAR Multichannel autoregressive
- MPAV Minimum peak amplitude value
- MSLT Multiple sleep latency test
- MT Movement time
- MTM Multi-taper method
- NREM Non-rapid eye movement
- OSA Obstructive sleep apnoea
- OSAHS Obstructive sleep apnea-hypopnea syndrome
- PLM Periodic limb movements
- PLMS Periodic leg movement syndrome
- PMT Power multi-taper

### PR Power rations

- PSD Power spectral density
- PSDR Power spectral density ratios
- PSQI Pittsburgh Sleep Quality Index
- R&K Rechtschaffen and Kales
- RBD REM sleep behavior disorder
- RBSE Relative band spectral energy
- REM Rapid eye movement
- RMSF Root mean square frequency
- ROC Right outer canthus
- RSE Relative spectral energy
- RSEB Relative spectral energy band
- SAHS Sleep apnea-hypopnea syndrome
- SE Spectral entropy
- SEF Spectral edge frequency
- SEM Slow eye movement
- SFS Sequential feature selection
- SHAS Sleep hypopnea -apnea syndrome

### SL Sleep latency

- SREM Sleep rapid eye movement
- SS Sleep spindles
- SWA Slow wave activity
- SWS Slow wave sleep
- SWSD Slow wave sleep duration
- TST Total sleep time
- WASO Wake time after sleep onset

# 1 : Introduction

### **1.1 Introduction**

Scientists have indicated that sleep covers approximately one third of a human being's life. Sleep refers to a period of consolidation of memory and brain recovery. In almost 1000 years ago, scientists wrote about sleepiness and sleep, indicating that the regulation of sleep is associated with heat resulted by human body [1]. Since this period, many scholars have offered different opinions in relation to sleep [2]. It has been pointed out in several studies that sleep is critical for energy preservation, for the restoration of biosynthesis (a simple structure converted to a complex structure in living organisms), facilitating learning and memory. When an individual is sleeping, the mind is strengthened, and motor functioning and performance are improved [3, 4]. It has been suggested that sleep is linked to metabolic function and obesity [5]. Furthermore, short and fragmented sleep has been associated with an increase in susceptibility to the common cold [6]. Sleep can impacts negatively on aspects of cognition, like vigilant attention, and public health [7, 8]. Sleep restriction or disorders can lead to sleepiness and may result in the involuntary onset of sleep, (falling asleep) causing car/truck accidents [9, 10, 11]. Changed wake/sleep patterns can impact negatively on the performance on neuropsychological tests and also in shift work [9].

Over a third of a century ago, Rechtschaffen and Kales (R&K) introduced a sleep manual with rules to detect different stages of sleep, using overnight polysomnography (PSG) [12]. Around a decade ago the American Academy of Sleep Medicine (AASM) revised and expanded the rules to encompass not only the sleep stages but also other abnormalities, including respiratory and cardiac events [13]. Sleep has been monitored through PSG with electrooculogram (EOG), electromyography (EMG) and electroencephalogram (EEG) electrodes and different sensors. Sleep stages are scored or classified using central EEG, left and right EOG, and chin EMG [12]. The manual scoring of sleep is a subjective and time-consuming process; hence, there is a demand for comprehensive and better automatic techniques that can be applied more easily, and can be used in experimental and clinical, ambulatory research.

Feature extraction is an important tool in the development of scoring algorithms for the detection of sleep stages. Most previous scoring algorithms depended on computationally undemanding, time-domain analyses, such as period amplitude [14] or the interval histogram method [15]. Most of these algorithms use 20- or 30-second epochs for EEG signal sleep stage classification either according to the R&K or AASM rules [16, 17]. However, there is some debate within the sleep research community in connection with the R&K or AASM sleep stage rules [18]. These rules might be accurate only for healthy young subjects and use a tentative resolution based on 20- or 30-second epochs to identify the different sleep stages. Therefore, in this thesis the candidate has attempted to develop an algorithm for automatic sleep stage detection based on consecutive and non-consecutive, 6-second sub-epoch EEG. On the other hand, the scalp EEG electrodes placement may cause sleep disturbance due to using many electrodes during the recording. Therefore, in this thesis the candidate has attempted to utilise a single EEG electrode for automatic sleep stage detection.

This thesis attempts to address the following sets of research questions. The first set focuses on sleep stage detection as follows:

1. Which of the PSG signals are the most efficient in the detection of the different sleep stages?

- 2. Can the EOG signal be used to automatically detect the sleep stages as an alternative to the EEG signal?
- 3. What feature extraction parameters from the PSG signals are most suitable to distinguish and characterise various sleep stages?

The second set of research questions focus on the sleep disorder classification as follows:

- 1. Are there significant differences in the PSG signals characterising healthy subjects and patients with various sleep disorders?
- 2. Can these differences be utilised to develop a screening method that can effectively classify various sleep disorders?

### 1.2 Motivation

In Australia, the prevalence of sleep disorders has been studied over different age groups in the general population [19]. In 2010, approximately 1.5 million people (8.9% of the population in Australia) were diagnosed with a sleeping disorder that included 199,000 people with restless limb syndrome (RLS) (1.2%), 492,000 with primary insomnia (3%) and 775,000 in the obstructive sleep apnoea (OSA) (4.7%) [12]. The direct annual cost, including health care and medication is estimated to be \$10 billion. The indirect costs (i.e. increased traffic accidents and work, reduced productivity and other medical illness vulnerability) can reach \$100 billion [19].

Various theories have been proposed to explain the physiological function of sleep [3]. However, no universally accepted theory exists. Therefore, for individuals to comprehend their sleep function, disorders and physiology, extensive sleep-monitoring studies should be undertaken on different subjects for an extended duration. This necessitates objective evaluation methods for the sleep process. The recording of sleep stages is vital to the clinical treatment and analysis of sleep disorders. The information obtained from the sleep stages is significant because it can be utilised in estimating respiratory disturbances in various stages of sleep.

### **1.3 Research aims**

The research aims can best be presented in the following manner:

- 1. Develop an automatic sleep stages detection method:
  - a. By utilising only EEG signal for the detection of all sleep stages based on the:
    - i. consecutive or non-consecutive EEG sub-epoch approach;
    - ii. evaluation of the most suitable spectral estimation methods (nonparametric or parametric);
    - iii. evaluation of the most effective consecutive or/and nonconsecutive EEG sub-epoch approach with different sleep disorders and ages;
  - b. By utilising only the EOG signal for the detection of all sleep stages based on the:
    - i. most suitable EEG and/or EOG feature extractions related to the sleep stages.
- 2. Develop an automatic classification of sleep disorders:
  - By utilising sleep architecture parameters and rules based on PSG sleep stage parameters;

b. Investigating the significant difference in architecture parameters between healthy and non-healthy sleep.

### **1.4 Objectives**

The general objective of this research is two-fold: firstly, to develop an automatic sleep stage detection method based on the EEG and/or EOG signals; secondly, to utilise the PSG features of sleep stages to develop an automatic sleep disorders classification system. The advantage of an automatic sleep stage detection method based on EEG or EOG signals can be utilised with portable sleep stage recording instead of using multichannel PSG signal. Classification of sleep disorders based on an automatic system can be improved with screening or diagnostic procedures which can be a faster and easier method.

### **1.5 Thesis composition**

This thesis is organised in the following manner:

**Chapter 1** presented an introduction to the research. This preamble included the motivation, research approach, and objectives.

**Chapter 2** gives an introduction to sleep and its basic concepts. Polysomnography and sleep variables are defined as the tools for the detection of the sleep stages. This chapter describes the difference between the R&K and AASM rules. An overview of automatic sleep stage detection and a description of its components is also presented.

**Chapter 3** describes the data and methodology used for the development and validation of the algorithm.

Chapter 4 describes the methods of automatic sleep stage detection.

Chapter 5 describes the classification method for the sleep disorders.

Chapter 6 presents in detail the results of the validation of the algorithms.

**Chapter 7** discusses the significance and limitations of the algorithms.

Chapter 8 presents a summary of the research contributions and also provides future works.

# 2 : Literature Review

#### 2.1 Basis of Sleep

The sleep phenomenon has gained reasonable scientific interest for an extended time. Sleep refers to a behavioral state that varies from wakefulness by a loss of reactivity readily and reversible in relation to events within one's environment [20]. The reversibility of sleep differentiates it from other types of states of consciousness, such as altered states of consciousness like the state of anesthesia or coma characterised by unresponsiveness and others. Different theories offer insight about the reasons that people sleep. According to the passive theory, sleep occurs because of a lack of sensory stimulation or to prevent tiredness [3, 21]. For many years sleep was considered as a passive state of the brain and the opposite of wakefulness [22]. It was assumed that the excitatory regions of the brainstem and other areas of the brain get exhausted and turn inactive. Hence, sleep was brought about by this inactiveness. The active theories also indicate that the brain aggressively deters consciousness [22].

Prior to the neurophysiologic analysis of sleep, the depth of sleep was assessed behaviourally [23]. Applying data from 211 nights, Michelson attested earlier documentation that sleep depth (estimated as arousal threshold) reached its maximum within approximately one hour after an individual fell asleep. In the course of the night, several sleep depths were indicated as many as four maxima and minima before the lowest depth of sleep during the morning [23]. Deep sleep was later interrelated to 0.5-3Hz low brain waves [24]. The effects of sleep deprivation were measured on several consequences, counting heart rate, memory, and urine examination [25]. Loomis et al. [26] undertook the first organized and systematic neurophysiologic recordings of sleep.

### 2.1.1 Non-Rapid Eye Moment Sleep

NREM is the classifications of deep sleep N3 (S3, S4) and light sleep N1, N2 (S1, S2). Deep sleep is also known as slow wave sleep (SWS). Sleep stage N2 (S2) is featured by K complexes (KC) and sleep spindles (SS). The abbreviations, MT (movement time), W, S1, S2, S3, S4 and SREM (sleep rapid eye movement), are obtained from the old standard of R&K (1968) [12] and R, W, N1, N2, N3 are derived from the standard of the AASM [13] (the difference between these two standards will explained in section 2.3).

An intensification of NREM sleep, especially during SWS, has been linked to recovery from a deficiency of sleep [27]. In NREM sleep, depolarization of cortical neurons occurs and fire tonically (the same as quiet wakefulness). However, the depolarization upstates are interrupted by short, hyperpolarized downs-states while the neurons stay silent [28].

The SWS part of NREM sleep is fundamental of the markers of the regulation of sleep [27]. Studies demonstrated that extended wakefulness led to an increase in the quantity of SWS in the subsequent recovery of sleep [29]. An EEG delta (0.5-4 Hz) power is applied instead of the visually scored SWS. This can be known as slow wave activity (SWA). Alteration in SWA and SWS have been found in various experimental paradigms [30]. After a total sleep deficiency, recovery sleep indicates an increase in the quantity of SWS [31].

### 2.1.2 Rapid Eye Movement Sleep

Approximately 140 years ago, Griesinger's studies suggested a link between somatic muscle and twitching of the eyelids during sleep [32]. In 1877, Dewar recorded eye movements using an electrical method and it was found that the electrical potential

occurred due to involuntary activities of the eyeball [33]. The early assessments presumed that the recorded potential included action-potential (the electrical membrane potential of a muscle rapidly increases and decreases) emerging from the ocular muscle [34]. Later, it was established that the recorded activity was due to the corneo-retinal potential (which is the remaining potential among the cornea and retina) [35]. However, it took long studies (until 1953) when Aserinsky and Kleitman discovered the rapid eye movement (REM) in the course of sleep [36].

A growing interest emerged in regards to assessing eye movement during sleep after Kleitman and Aserinsky's work. The density of REM (DREM) and REM sleep was linked to sleep, essentially after 1969 [36]. According to Aserinsky, DREM occurs almost after seven to ten hours of sleep [36]. The density of eye movement reduces during recovery sleep after the occurrence of sleep deprivation [37]. Lucidi and colleagues found that a reduction in the density of REM parallels an upsurge in SWS [38]. De Gannaro and colleagues also reported on the similar results [39].

The density of eye movement is increasing in REM sleep periods followed by periods of awakening then followed by NREM sleep [40]. The density of eye movement has been recorded to be highest in 5-10 minutes after the beginning of the REM sleep period followed by a significant reduction after 10 minutes [41]. The density of eye movement has approximately a 10-minute periodicity [42].

### 2.2 Sleep disorders and psychiatric disorder

Sleep disorder refers to a medical condition, also known as somnipathy [43]. Contemporary studies have argued that severe sleep disorders can interfere with the normal mental, social, emotional and physical functioning of an individual [44]. Actigraphy (a non-invasive technique for monitoring human activity period) and polysomnography (PSG) (more information in section 2.4) are some of the common tests that have been used in the diagnosis of sleep disorders.

Lack of quality sleep has been associated with adverse impacts on emotional balance, energy, and health. People with good health usually tend to have quality sleep while those suffering from repeated disorders of sleeping might have an underlying severe or minor mental or medical problem. Evidence indicates that even minimal sleep loss puts a toll on people's moods, wellbeing and their ability to cope with stress [44].

### 2.2.1 Primary insomnia

Subcategory insomnia complaints commonly include difficulty in initiating (sleep onset insomnia) or/and maintaining sleep. They include extended periods of sleep onset insomnia or/and maintaining insomnia amounts of nighttime sleep [45]. The diagnostic and symptom category of insomnia are best denoted by their subcategory. These subcategories are described by different combinations of repeated sleep problems with sleep duration, initiation, quality and impairment during the daytime [43]. Insomnia complaints can be associated with the perception of non-restorative or poor quality sleep even if the quantity and quality of sleep episodes are perceived as adequate or regular. The meaning of insomnia being a complaint of sleep maintenance, sleep initiation, non-restorative sleep or associated with daytime impairment [46].

Primary insomnia can comprise both extrinsic (the body effected by outside factors) and intrinsic (the body effected by inside factors) factors involved in their etiology. They are not regarded as being secondary to another disorder [43]. The National Institutes of Health Consensus Development Conference for Insomnia in 2004 led to a promotion of comorbid insomnia term to differentiate primary insomnia from forms due to other sleep disorders, psychiatric and medical disorders and insomnia due to drug or medication use [43, 47]. The term, primary insomnia, is applied in the Diagnostic

and Statistical Manual of Mental Disorder (DSMMD) and is listed in the International Classification of Disease (ICD) [48, 49, 50]. Hence, it has a significant advantages for use as a global categorization of insomnia.

The ICSD defines primary insomnia as a syndrome consisting of paradoxical insomnia, psychophysiological insomnia, and idiopathic insomnia [51]. Primary insomnia is regarded as a difficulty in maintaining sleep (early morning awakening or mid-sleep awakening), a difficulty in initiating sleep (sleep onset insomnia) or even non-restorative sleep which is chronic and persists for more than three weeks, notwithstanding that one has adequate opportunity for rest/sleep and it impairs daytime performance [52,53].

Many abnormalities (see Table 1) have been reported with primary insomnia patients [54]. These can be measured using PSG, with a particular focus on the REM stage and N1 stage. It has been reported that patients with primary insomnia most often suffer from difficulty falling asleep as well as intermittent wakefulness during sleep [53]. Moreover, the PSG characteristics in primary insomnia patients show an increase in N1 and a decrease in sleep efficiency (SE) [53]. Table 1 describes the PSG features of primary insomnia disorders.

Increase in N1 sleep stage
Increase in Sleep latency (SL)
Increase in REM sleep
Decrease in total sleep time (TST)
Increase in N1 sleep stage
Decrease in sleep efficiency (SE)

Table 1: Polysomnographic features for primary insomnia disorders

### 2.2.2 Periodic limb movement syndrome

The periodic limb movement syndrome (PLMs) is when a person is unaware of the limb movements caused by muscle contractions during sleep [1, 55]. The episodic limb movement condition occurs during sleep as an independent disorder of highly stereotyped and repetitive limb movements [56]. It is often associated with the syndrome of restless legs [56, 57]. Abnormal considerations include more than five involuntarily movements per hour in sleep whereby each limb movement lasts between 0.5 to 5 seconds [55, 58].

Periodic leg movement disorder is associated with a dopamine (in the brain it is a function as neurotransmitter) responsive disorder, and can be improved through dopaminergic agonistic management [50]. It has been reported that PLMs occur at night during the interval before a person enters REM sleep [50]. According to ICSD that PLMS occur in N1 of the sleep period before reaching REM sleep [50]. Moreover, during the REM period of sleep, an individual's voluntary muscles are paralysed, keeping them from acting out their dreams. The characteristics of PSG of PLMs are summaried in Table 2.

	Increase in N1 sleep stage
PSG features of PLMS	Decrease in N2 and N3 sleep stage

Table 2: Polysomnographic features of PLMs syndrome

### 2.2.3 Obstructive sleep apnea-hypopnea syndrome

Obstructive sleep apnea-hypopnea syndrome (OSAHS) has increased in the last 50 years, causing significant mortality and morbidity in both the developing and developed countries [59]. It causes daytime sleepiness [60]. Hypopnea and apnea are reasoned by

Absent during REM sleep

the airway being blacked during inspiration at sleeping periods. This condition occurs when the upper airway dilating muscles (striated muscles) normally relax when sleeping [59]. Patients with OSAHS have no ability for their dilating muscles to oppose the negative pressure in the airway in the course of inspiration [59].

Moreover, most patients have narrow upper airways. The upper airway patency is kept by dilating muscles and it has a higher activity during awake patients. However, in the course of sleep, the muscle tone decreases, narrowing the airway [61, 62, 63]. This may cause snoring with a subsequent airway obstruction and the apnea is episodes. The features of this case are hypercapnia, hypoxemia, changes in the large intra-thoracic pressure (120 mm Hg) and increases in systemic pressure of blood up to 250/150 mm Hg, which is linked to sleep fragmentation and sleep arousal of up to 100 times per hour [59, 61, 62, 63, 64].

The consequences OSAHS include social and neurobehavioral differences, such as: impaired vigilance, excessive daytime sleepiness, cognitive dysfunction, and mood disturbances. Sleepiness can lead to the inability to work efficiently, prevents socialising and can ruin interpersonal relationships [61]. Current studies have demonstrate that sleep-related disorders such as obstructive sleep apnoea may increase the risk of stroke or death [65]. The clinical examination is limited and not sufficiently sensitive; therefore, the PSG features are very important and required for screening the Obstructive sleep apnea (OSA) patients. For example, it has been reported that the characteristic of PSG with OSA patients during sleep is shown increase in sleep N1 and N2 stage, as shown in Table 3 [1].
1 able 3: Polysomnographic features of Obstructive sleep apnea	Table 3: Pol	vsomnographic features of Obstructiv	ve sleep apnea
--	--------------	--------------------------------------	----------------

	Increase in N1 and N2 sleep stage
PSG features of	Decrease in N3 sleep stage
OSA	Decrease in REM sleep stage
	Increase of frequency of arousals

#### 2.2.4 Schizophrenia

Schizophrenia is a sleep mental disorder, with this ratio of one in one hundred people that are diagnosed [66]. This disorder affects human on psychiatric perceptions, emotions behavior and thoughts [66]. Sleep problems in individuals with schizophrenia are a familiar occurrence [66]. Although there is no evidence that sleep disorders lead to psychiatric disorders, there is a link between sleep and psychiatric disorders, like depression and schizophrenia, with studies indicating that psychiatric disorders are the leading cause of sleep disorders [67]. Disturbed sleeping patterns are often found in 30-80% of patients with schizophrenia, and it mainly depends on the extent of the psychotic symptomatology [68, 69]. According PSG analyses, total sleep time and sleep efficiency typically decrease, while increased latency of sleep are found for patients with schizophrenia, as shown in Table 4. Studies have reported alterations in N2 sleep stage, REM and SWS, with decreasing REM and REM latency [68, 70]. Some evidence has indicated that patients with schizophrenia have a higher risk of experiencing sleeplinked breathing disorders, more especially the groups who are overweight and have a long-term use of antipsychotics [68]. It is unclear whether restless legs syndrome or periodic leg movement during sleep are found in lower or higher occurrence in schizophrenic individuals [68]. There is no consistent impact of the first-generation antipsychotic medication on sleep continuity, as well as a measure of sleep structure, including REM latency and sleep stages [71, 72].

	Increase in SL
DSC factures of achigophronic	Decrease in SE Increase in wake time after sleep
PSO features of schizophrenia	onset (WASO)
	Decrease in N3 sleep stage
	Decrease in REM sleep

Table 4: Polysomnographic features of schizophrenia disorders

#### 2.2.5 Classification of sleep disorders

The classification of sleep disorders is essential to differentiate between disorders and enhance the understanding of etiology (that is used to study the cause of sleep disorder phenomena), pathophysiology and symptoms that can enable appropriate treatment [42, 73]. The earliest systems of classification primarily distinguished between sleep disorders depending on the main symptoms and abnormal events occurring in sleep, excessive sleep and insomnia. Hence, the earliest systems of classification were not based on the pathophysiology of sleep since the causes of most disorders were unknown [42, 74]. The first noteworthy classification of sleep disorders was published as the 'Diagnostic Classification of Sleep and Arousal Disorders' [75]. In 1993, a significant effort was made to publish the International Classification of Sleep Disorders (ICSD) as a collaboration between different international sleep societies, such as the American Sleep Disorders Association (ASDA), the European Sleep Research Society, the Japanese Society of Sleep Research and the Latin American Sleep Society [76]. The ICSD has been extensively utilised by clinicians and is essential for diagnostic, epidemiologic (which deals with the study of the patterns, causes and the effective control of disease in the population) and research purposes.

The ICSD version 2 (ICSD-2) was disseminated in 2005, and is currently undergoing further amendments [50]. It integrates symptomatic presentation, such as insomnia with one structured in part of the pathophysiology (i.e. circadian rhythms [changes in mental, physical and behaviour during a 24-hour period]) and body systems (i.e breathing disorders). This systematic structuring of sleep disorders is crucial since the pathophysiology of various disorders is still not well-known. The ICSD-2 offers relevant epidemiological and diagnostic knowledge on sleep disorders for easier differentiation from the other emerging disorders [50]. According to the ICSD, there are more than 80 types of sleep-related disorders; among the most common are sleep apnoea, snoring, insomnia, circadian rhythm and parasomnias [77].

#### 2.3 Visual sleep stage scoring

The quantitative visual sleep stage scoring includes sleep stage or event epoch marking, analysis and counting [13, 78]. Events include eye movements, K complexes and sleep spindles. A standard approach to visual analysis includes the segmentation of sleep into 20 or 30 second epochs of movement time (MT), wakefulness (W), sleep stages S1, S2, S3 and S4 from EOG, EMG and EEG .With visual scoring, this data is required to offer consistent, practical and quantitative sleep structure or architecture information [79]. Determining the sleep stages is crucial in the clinical treatment and the diagnosis of sleep disorders [80].

Alongside sleep staging, polysomnography comprises simultaneous reports of various other sleep signals, such as cardiac activity (ECG), blood oxygenation (oxygen saturation - SpO2), and respiration (thermistor, thermocouple). These signals serve as vital measures in the clinical assessment for sleep disorders, as depicted in the ICSD manual [76]. As indicated by Schulz [81], visual sleep staging has undergone an

evolution that led to a standardisation of the rules by Loomis in the 1930s [26], Aserinsky and Kleitman in 1950s [35] and by Rechtschaffen and Kales in 1960s [12].

#### 2.3.1 The Rechtscaffen and Kales scoring rules

Recognising the importance of the ever-increasing need for standardising sleep scoring, a committee was formed comprising the members of the sleep research society in order to create a standardised scoring manual [82]. Rechtschaffen and Kales (R&K) formed standard criteria that segmented sleep into the sleep stages of REM S1, S2, S3, and S4, wakefulness and movement time (MT) based on characteristics of the EMG, EOG and EEG [12, 83]. The SWS is visually defined as waves of lower frequency (below 2Hz), having an amplitude greater than 75  $\mu$ V (peak to peak). An epoch is described as stage S3 when at least 20% of the epoch (20 seconds) includes SWS and is scored as S4 when more than 50% of the epoch (20 seconds) includes SWS. Stages S4 and S3 are known as SWS and the amplitude is measured from the C3-M2 or C4-M1 channels. M2 and M1 are mastoids, and are used as reference points to measure EEG potentials. Any epoch that does not satisfy S4 or S3 criteria belongs to another sleep stage [84].

The S1 stage of sleep is described as relatively low voltage and mixed-frequency EEG with an activity prominence within the range of 2 to 7Hz. The vertex sharp waves from EEG are rarely as high as 200  $\mu$ V appear in S1. Also, the S1 stage can be described by slow eye movements. When alpha activity is less than 50% of the epoch and with relatively mixed-frequency, and low-voltage is at least 50% of the epoch, the epoch is classified as stage S1. Sleep spindles or K complexes denote stage S2. Also, S2 is classified when a time interval is less than 3 minutes between sleep spindles and/or K complexes without a sign of pronounced increase or movement arousal of the muscle tone. The REM is relatively low-voltage, mixed-frequency EEG in connection with low

amplitude EMG and episodic REM [12, 84]. Moreover, detailed rules exist for the offset and onset of REM. Fundamentally, REM duration is continuously scored as an REM stage until muscle tones are amplified or spindles appear.

#### 2.3.2 The American Academy of Sleep Medicine scoring rules

The American Academy of Sleep Medicine (AASM) scoring rules represent the development and evolution of the 2007 manual with an objective of consolidating, standardising and restructuring the original (R&K) sleep staging standard rules [13,85]. A major revision to the earlier scoring included changes to EEG electrodes placement of the frontal position to the typically used occipital and central position [13, 81]. Another change included the reclassification of sleep stages to stages N1 to N3 (non-REM),W and R for REM stages [80]. The alteration of abbreviations was to prevent confusion between the two standards of classification whereby the stage S3 and S4 (in R&K rules) were joined to form stage N3, since no clinical or physiologic basis had been discovered for a difference between stages S4 and S3 [51, 86].

Other notable changes included an alteration to the PSG data analysis, which has potential clinical effects [13]. For instance, large variance in the indexes of adult apneahypopnea when using the recommended AASM rules versus the earlier rules ("if there was > 50% airflow reduction alone or a lesser airflow reduction associated with  $\geq$  3% desaturation or  $\geq$  3-sec EEG arousal"), a decrease in sleep N2 with an increase in sleep N1, and transitions of sleep due to a rule prevailing transition from sleep N2 to N1, increased sleep N3 in adults with the addition of the EEG frontal electrode, enhanced slow wave detection and enhanced reliability of inter-scorer [87].

#### 2.4 Polysomnography and Hypnogram

Polysomnography (PSG) assesses comprehensive sleep using numerous electrophysiological signals. It is the gold standard for measuring sleep stages [88]. Also it identifies stage-dependent pathologies and determines severity and cause [89]. The PSG studies are utilised by clinicians to analyse sleep quality in individuals who have sleep disorders or in research studies for both patients and normal subjects [90]. The PSG can be recorded in a laboratory or at home. It can also investigate the effectiveness of therapeutic regimens. In 1997, the AASM issued practice parameters or signals for PSG and related processes, including cardiorespiratory and oximetry studies that were later updated in 2005 [91].

In the course of PSG studies, numerous electrodes were placed on the individual to measure activities of the eye, brain, heart, and muscles movement [92]. These signals were interpreted by sleep technologists to create an extensive sleep report of the individual's night sleep. Table 5 shows some of the recommended parameters or signals to be reported for PSG, such as sleep scoring data, and arousal, respiration, cardiac and movement events according to the AASM standards [13].

Parameter	Sleep scoring	Arousal	Respiratory	Cardiac	Movement
	data	events	events	events	events
EEG and	Light out clock	Number	Number of	Average	Number of
EOG	time (hr: min)	of arousal	obstructive	heart rate	PLMS
derivation			apneas	during sleep	
Chin EMG	Total Sleep Time	Arousal	Number of	Highest	Number of
	(TST; in min)	index	mixed apneas	heart rate	with arousals
				during sleep	
Leg EMG	Total recording		Number of	Highest	PLMS index
derivation	time (light out to		central apneas	heart rate	
	light in min)			during	
Oxygen	Percentage of SE		Number of	recording	PLMS
Saturation			hypopneas		arousal index
Airflow	Time in each		Hypopnea index		
parameters	stage (min)				
Body	Percentage of		Respiratory		
Position	TST in each		effort linked to		
	stage		arousal		

Table 5: The recommend parameters to be reported for PSG during sleep [13].

The AASM has developed different recommendations for PSG for various sleep disorders [91]. The PSG with extended video and EEG are recommended for violent sleep behaviors, sleep walking, REM sleep behavior disorders (RBD: patients physically act out their dreams), and forensic cases [91]. For cases of parasomnia (undesirable behavioural or experiential events associated with sleep), PSG is unusual. In cases of narcolepsy (condition characterised by extreme tendency to daytime drowsiness), PSG is combined with the multiple sleep latency test (MSLT, a diagnostic tool for sleep disorder) that should be performed the following day. In nocturnal seizures, PSG with extended tibialis EMG, video and EEG are recommended [91].

PSG requires considerable capital investment in health-care, dedicated, well-trained technical staff and accurate bed space. The interpretation of the generated PSG data is a time-consuming process overall, with the cost of undertaking signals overnight PSG assessment or recording estimated to range between \$1,000 and \$2,000 [93].

A hypnogram is sleep architecture that represents a form of polysomnography or a graph that depicts the stages of sleep in relation to time. A hypnogram allows for the identification of different stages of sleep, including NREM and REM. Typically, a hypnogram derived from night-long sleep recordings can graphically summarise the stages of sleep a person experiences. The hypnogram usually indicates the descent from W to stage N1 (lasting between 1 and 7 min), N2 (the duration between 10 and 25 min), N3 (can lasting up to 40 min) and REM (can lasting up to 90 min) [12, 94]. The duration of the cycle of sleep stage NREM-REM is expected to be between 90 and 120 minutes, and might be repeated between 4 to 6 times per night. Figure 1 shows a hypnogram sample with the sleep stage and its time duration.



Figure 1: Hypnogram sample, where the y-axis indicates the sleep stages and the x-axis indicates time (epoch) [95].

#### 2.4.1 Electroencephalography (EEG)

EEG measures and spontaneously records the electrical activity of the brain from the scalp [96]. It evaluates voltage potentials resulting from the ionic flows of current within the brain neurons. EEG includes multiple electrodes placed on the scalp [96].

The brain waves generated by the EEG system comprise various frequency bands, such as delta, alpha, theta and beta wave bands [97, 98], as shown in Figure 2. Delta waves range from 0.5 to 4 Hz frequency with 20 to 400  $\mu$ V amplitude, and occur during low brain activity, such as medium anaesthetic state and deep sleep [99]. Alpha waves range from 8 to 13 Hz frequency with a 2 to 10  $\mu$ V amplitude range. They are usually recorded when a person is awake with closed eyes and in a mental and physical state of rest. Beta waves are recorded at higher frequencies, ranging from 13 to 30Hz. Their amplitudes range from 1-5  $\mu$ V [100]. Beta waves are observed at concentrated attentions during the mental working state and also during the REM stage of sleep.



Figure 2: EEG frequency bands, where the y-axis indicates EEG frequency bands and the x-axis indicates time per second (sec) of the brain [101].

EEG signals can differentiate the stages of sleep. There are five phases observed in one's NREM sleep. The Phase 0 wakefulness stage is characterised by low amplitude EEG levels and has two frequency bands – an alpha band of 8–13 Hz and a beta band of 13–30 Hz. Alpha waves are hallmarks of this stage. The Phase 1 stage, associated with drowsiness, is distinguished by higher EEG amplitude levels than the wakefulness stage, and theta bands of 4–7 Hz are more dominant. Two important events, known as the sleep spindle and the k-complex, occur during the Phase 2. The sleep spindle event has a frequency range of 12–16 Hz and can also be present in the Phase 3. Slow wave sleep (SWS), one of the hallmarks of Phases 3 and 4, has a lower frequency range of 0.5–2 Hz but also exhibits higher amplitude EEG levels than the other stages [98]. Table 6 shows the NREM sleep phases and their features.

NREM Sleep Phases	Characteristics
Phase 0 (wakefulness)	Low-amplitude EEG, alpha rhythm (8–13 Hz) prominent over occipital regions.
Phase 1	Low-amplitude mixed-frequency and low- amplitude EEG pattern, with theta (4- 7 Hz) rhythm.
Phase 2	EEG with one or more non-arousal KCs or one or more trains of SSs.
Phase 3 and 4	Low-frequency deep sleep EEG activity.

Table 6: NREM sleep phases and their characteristics [95].

REM sleep refers to the dream state and dreams experienced in this stage can be remembered while in the wakefulness state [101, 102]. This REM stage is usually interspersed between the other stages of sleep, and is linked with low-amplitude EEG, deprived of KCs or SSs, REMs and low chin EMG tone activity [96]

2.4.1.1 10-20 EEG electrode placement system

10-20 system of EEG electrode placement refers to the technique used to depict the location of the scalp electrodes [103, 104,105]. Scalp electrodes are applied in recording EEG through a machine known as an electroencephalograph. The EEG brain activities are generated from activities of thousands of neurons within the brain. The variation of patterns demonstrates the changes of an individual's state. For instance, a relaxed state shows slow EEG waves (8 to 13 Hz) and a state of arousal shows faster waves (13 to 30 Hz) [105].

The 10-20 system is usually applied to describe the scalp electrode positioning of EEG recording [105, 106]. The objective of a 10-20 system is to standardise electrode positions or locations, with the numbers, 10 and 20 describing the distances between adjacent electrodes which are set at 10% or 20% of the right-left or front-back of the skull distance [105].

The 10-20 system is based on the connection between the location of a given conductor and the fundamental area of the cerebral cortex [107]. Each brain area has a letter for the lobe and hemisphere location. The letter O, P, C, T and F stand for the occipital, parietal, central, temporal and frontal respectively [108]. For example, the letter, C, represents the central lobe, z (zero), which is the electrode placed on the midline, and the even numbers (2, 4, 6 and 8) denote electrode positions within the right hemisphere whilst the odd numbers (1, 3, 5 and 7) denote the left hemisphere [105, 107]. Figure 3 shows the electrode positions of the 10-20 system.



Figure 3: The EEG electrode positions of the 10-20 system [101].

#### 2.4.2 Electrooculogram

Electrooculography (EOG) is a method that records the cornea-retinal standing potential existing between the front and back of the human eye [108]. The resultant signal is known as the electrooculogram. In measuring the eye movement, electrode pairs are located either below and above or to the right and left of the eye [108]. An EOG signal response of positive or negative defection can be recorded if the movement of the eye is made from the central position towards one of the electrodes. Therefore, a potential difference would occur when the eye moves between the placed electrodes. The eye has a standing electrical potential known as the corneo-fundal [109]. This potential is decreased during darkness.

The earlier studies had a different EOG positioning of the electrodes. Loomis and colleagues [110] used only the referential electrode placed above the left eye whilst

Aserinsky and Kleitman recorded two bipolar channels on a single eye [35]. Later, Hord incorporated four electrodes into one bipolar channel in order to increase the common mode rejection ratio of an EEG system (amplifier) [111] but this positioning had shortcomings in the analysis of automatic sleep [112,113].

In the standard manual of the AASM, the EOG electrode positioning is recommended as approximately 1 cm above and faintly lateral to the outer canthus of one eye (E1 and E2, as shows in Figure 4) and a reference electrode on either the mastoid or homolateral ear lobe. Figure 4 shows three different electrode positions.



Figure 4: Recommended, previous and alternate electrode locations of the EOG signal. LOC = left outer canthus; ROC = right outer canthus [95].

#### 2.4.2.1 Slow and Rapid Eye Movements

By 1929, Miles had recorded the importance of the movements of the eye in the changeover from wakefulness to sleep [114, 115]. The difference between drowsiness and alertness was evident in analysing the behavior of the eyes. It is documented that the horizontal eye movements observed at sleep onset were similar to the pursuit movements when an eye is closely watching an object such as a slow swaying swing [115]. According to the manual sleep scoring rules of R&K, the slow eye movements (SEM) do not determine the onset of sleep but are linked to sleep onset, especially S1

which follows wakefulness, is featured by the SEM each taking several seconds, prominent at the early portions of the S1. Within the new scoring (AASM), the SEMs are used to define sleep onset for individuals who do not produce alpha rhythm [13]. They are described as "conjugate with a reasonably regular and sinusoidal eye movements with the initial deflection usually lasting >500 ms" [13]. Other scholars have defined SEM to be 0.25 Hz horizontal, pendular eye movement [116,117]. With the EOG channel, SEM is described as horizontal and lasting for 1 sec or longer at 100  $\mu$ V amplitude. SEM can be recorded by utilising an electrode on the eyelid [116,118]. This nightcap system employs a 25 mm x 7 mm piezoelectric film that is adhesive-backed and placed on the upper eyelid [118, 119]. The SEM has also been defined as [120]:

- slow sinusoidal excursion of 0.2 to 0.6 and lasts more than 1 sec;
- absence of artefacts like EEG/EMG and blinks;
- movements beginning and ending at approximately zero velocity;
- onset of left and right movement occurs in 300ms of another;
- binocular synchrony with opposed-phase detections in the two EOG channels;
- amplitude of 20-200  $\mu$ V.

For REM (i.e saccades), many automatic systems for detection have been applied [121, 122]. Many cognitive elements are probed with the movements of the eye, such as responsiveness [121]. With digital progression, Haddad and Gopal created a system based on an amplitude and slope of the detection of eye movement in REM sleep [122]. Tsuji et al [123] utilised a wavelet transformation whilst Tan el al [124] used period–amplitude and fast fourier transform (FFT) analysis to count the number of eye movements during REM sleep. Two-channel EOG was used, referenced to ipsilateral

mastoids. Signals were filtered using a 4th order Butterworth bandpass filter at 1 to 5 Hz using the following steps [125]:

- a) **Candidate REM detection**: produces instantaneous signal of at least threshold 10  $(\mu V)^2$  with a local maximum if no higher peak within a time window of 1 sec.
- b) **Detection REM features**: Maximum absolute amplitude of right or left EOG, correlation coefficient of two EOG, Calculation of the negative instantaneous product, deflation angle (angles utilising 0.2 sec of data on the right and left side of the peak)
- c) **The deflection angles rule for detection REM**: If angle for both the right and left more than 45 degree or if the angle of left side more than 30 degree and the right side more than of 60 degree or if the angle of left side more than 60 degree and the right side more than of 30 degree.

#### 2.5 Automatic sleep stage detection

The development of the first automated EEG analysis system applied to sleep recording [126]. Drohocki's system was employed in plotting EEG integrated amplitude during the night [127]. The highest values were recorded during S4 and S3. Another researcher employed bandpass filtering, detecting rhythmical EEG wave band [16]. Frost developed portable analog sleep detecting machine that uses 1 channel EOG and 1 channel EEG [128]. This system utilised 0.7-13 Hz amplitude EEG filter and period to produce a continuous sleep depth curve that has the threshold to detect the sleep stages. Three multiple level amplitudes of 20%, 1% and 100% were used, whereby 100% corresponded to wakefulness (highest amplitudes) and only the peaks were integrated with periods. The periods were detected when 20% and 1% thresholds were crossed in a given sequence. The lowest curve values had low periods and high counts of high

peaks. The indicator was 2 to 3 Hz EOG amplitude of REM if EEG showed S1 sleep. Several systems used period–amplitude analysis [129], and analog, hybrid, and early digital automatic sleep analysis systems were reviewed by Hasan [112].

The main purpose of the automated sleep detection and monitoring systems is to accurately detect the sleep stages and the microstructure (i.e. sleep arousal) [130, 131, and 132]. As mentioned, earlier sleep staging is based on the impression that the pattern (characteristic EEG patterns) will exist for a given interval of time before a new pattern emerges, which indicates a change of stage. There is a continuum from sleep stage N2 to sleep stag N3. The artificial differentiation of sleep stages is a facilitation implemented to standardise the examination of automated detection across the sleep laboratories and reviewers [130]. The exact period of change of sleep stages is extremely subjective and leaves room for different interpretations of what the sleep stage indicate to the epochs by scorers. Studies have indicated inter-scorer agreement (epoch by epoch agreement) between 67% and 90% depending on the different number of readers and scoring epoch lengths [133]. Most data from the inter-scorer agreement is based on the study of normal or healthy subjects.

Automatic sleep analysis comprises few consecutive steps, as shown in Figure 5.



Figure 5: The automatic analysis process [130].

The pre-processing step involves the reduction by statistical tools of the vast amount of the generated raw data to be easily managed. It includes removal of the artefacts in the EEG signal [130, 135]. The next step, feature extraction, can be the power or amplitude

of regular EEG waves (i.e. delta, theta, alpha and beta bands) and transients, such as SS and KC [130]. The analysis can be undertaken in short epochs, ranging from 1 to 20 or 30 seconds to avoid the loss of time resolution [134]. The final step includes the combination of EEG signals and extracted features to reduce the number of sleep stages.

Rule-based logistics are utilised in classifying the stages based on features that are extracted from the EEG that imitate visual inspection according to standards, such as the AASM rules and R&K [130,135]. For clinical systems, the final outcomes of the automatic sleep scoring are required to closely imitate the visual AASM or R&K scoring. Most published studies on automatic sleep stage detection have produced outcomes from healthy persons with dependability between 70 and 90% [130,135].

Feature extraction is an important part in the development of sleep stage scoring algorithms. The majority of the previous sleep stage scoring algorithms depended on computationally undemanding time domain analyses, such as period amplitude [14] as well as interval histogram method [15]. Subsequently, most of the algorithms have been developed before/after 1990 by feature extraction methods, utilising the power spectrum of the EEG, EOG or EMG signals. There are several methods that can execute spectral decomposition by means of discrete Fourier transform (DFT) /FFT [136, 137], autoregressive modelling [138], adaptive filter banks or wavelets [17]. Today, innovative spectral analysis methods are being used for the EEG sleep staging, such as normalisation amplitude of frequency bands [139, 140] and features derived from harmonic, predictor coefficients and Hjorth sub-band energy parameters [141].

Most algorithms have utilised 30- or 20-second epochs for EEG sleep stage classification according to the R&K or AASM rules [17,142]. Because the EEGs are non-stationary signals, the tradition of sleep stage scoring based on a 30-second epoch

might provide less information about the brain activities because it uses a fixed epoch duration. It has been suggested that EEG epoch duration be reduced to 1, 2 or 5 seconds [143].

Due to the non-stationary nature of the EEG signal, time-frequency analysis is used in the decomposition process for classification. The matching pursuit, discrete wavelet transform, wavelet transform modulus maxima, and Gabor transform have been applied to detect the sleep stages [143, 144, 145, 146, 147].

Also significant work has been done in creating systems for sleep staging based on nonlinear and linear classifiers, including fuzzy logic, artificial neural network (ANN) and pattern recognition [81, 148,149]. These classifiers do not need complicated domain knowledge or classification rules. The ANN characterise by learning ability which lets them create their own structures, depending on training sets [130].

One of the primary approaches of computer-assisted sleep stage detection or scoring is the 'special purpose sleep analysing hybrid computer' which utilised three EEG and two EOG signals [142]. This study was able to analyse data in online mode, based on a linear analog filter for detection of sleep spindles as well as the alpha EEG signal. The assessment was made based on a study of 15 healthy subjects aged between 15 and 21 years. The inter-rater agreement between their system and the manual scoring was 83.5%. However, several studies indicated that it was easier to detect sleep stages in young, healthy subjects than in older subjects who suffered from sleep disorders [150]. Agarwal and Gotman [151] presented a computer-assisted sleep staging method that utilised the principles of segmentation and self-organisation (clustering) based on primitive, sleep-related features to find the pseudo-natural (cluster pattern) stages. Their method was developed and tested using 12 subjects of many types (healthy, abnormal [such as PLMS and sleep apnea], male, female, and different age groups), demonstrating an overall agreement of 80.6% with a manual scoring of 20-second epochs. However, this study reported that the maximum amount of errors occurred in the identification of the highly transitional S1, 54% of which were misclassified into neighbouring S2 or W. Tim Schluter et al. [152] developed automatic sleep stage scoring apnoea-hypopnoea detection based on EEG, EOG, and EMG signals. They utilised several combined techniques, such as FFT, wavelets, derivative dynamic time warping (DDTW), and waveform (pattern) recognition. Their approach was to extract features (frequencies and special patterns) from these signals and use the decision tree classifier to classify the sleep stages. The outcome of this study showed that the accuracy of sleep stage detection was 95.2%. However, this study used several databases without specification of the age or gender of all of its subjects.

Anderer et al. [150] presented automatic sleep stage detection and a classification based on 1 EEG signal, 2 EOG signals and a chin EMG channel. They utilised several methods for automatic identification of sleep/wake-related patterns based on the previous signals, such as to identify alpha and delta signals using period-amplitude analysis, model-based detection, band-pass filtering and spectral analysis. They also used the discrete wavelet transform and period-amplitude analysis methods to identify eye movement and maximal peak-to-peak amplitude as features extracted from the EMG signal. The classification system was based on the decision rules. The outcome of this study showed that the accuracy of a large database compared to manual scoring was 80%. Figure 6 compares the scoring of hypnograms by human experts with automated scoring by the commercial Somnolyzer  $24 \times 7$  system (which developed by Siesta Group in Austria).



Figure 6: The hypnogram scoring of different human experts vs the Somnolyzer  $24 \times 7$  system. The first hypnogram is indicate to the first expert scorer of sleep stage, the second hypnogram is the second expert scorer of sleep stage, the third hypnogram is indicate to the consensus scorer and the final hypnogram is indicate to the Somnolyzer  $24 \times 7$  system scoring [150].

However, this study faced obstacles in terms of the misclassification for some sleep and wake stages. In particular, subjects with sleep disorders, such as sleep apnea and insomnia had an accuracy of 75.6% and 85.5%, respectively. Susmakova et al. [153] utilised a spectral analysis and the fractal dimensions of the EEG, EOG, and EMG signals and obtained an accuracy of 77% in 20 healthy subjects. The authors faced obstacles in terms of distinguishing between stage N1 and REM sleep that eventually affected the overall accuracy. Recently, several studies have attempted to develop an automated sleep stage detection based on the EEG signals only [14,154]. Zhovna et al. [154] presented a novel method for automatic detection and classification of sleep stages using a multichannel autoregressive (MAR) model. The classification was performed using Kullback–Leibler (KL) divergence. The outcome of the study showed a classification accuracy of 93.2%. However, there was a weakness in their system in that it laced the W stage classification. Liu et al. [14] demonstrated a novel method based on the Hilbert-Huang transform (HHT) and back-propagation (BP) neural

network for automatic sleep stage detection and classification. The HHT was used as a feature extraction from the 30-second epoch of the EEG signal. The results revealed that the method was effective and promising in automatic sleep stage classification. However, this study was not able to differentiate between the N1 and REM stages; thus, they combined these two stages.

Another study employed 2-channel EOG for automatic sleep stage classification, EOG signal were referenced to the left mastoid (M1) [137]. The synchronous EEG activity SWS and S2 were identified by measuring peak-to-peak and the cross-correlation amplitude difference in the 0.5 to 6 Hz and between the 2-channel EOG. Automatic SEM approximation was utilised to designate W, S1, and REM. The EEG alpha power 8-12 Hz and beta power 18-30 Hz were used for detecting wakefulness [137]. Synchronous 1.5 to 6 Hz EEG activity and absence of great movements of the eye were used for the separation of S1 from REM. Also, simple smoothing rules were applied. The EEG, EMG and EOG of sleep were recorded for 256 subjects. To tune the system 132 training subjects' data were used and applied to 131 validation subjects that had different SWS, and S2 epochs were epoch-by-epoch agreement of 72% with Cohen's Kappa (CK) at 0.62. The results indicated an improvement of the specific EEG alpha thresholds for offline applications to 0.63 and 73% [137]. This method can further be developed in order to be applied in ambulatory sleep recording by employing only four disposable, self-applicable and self-adhesive electrodes. Figure 7 shows the study results.



Figure 7: Recording from group (a) percentages of slow wave (SW2 where 2 refers to rule number 2) segments; (b) SW3 (using rule 3) percentages; (c) percentages of rule S; (d) percentages of rule S; (e) automatic sleep stage scoring developed; (f) visual sleep scoring.

One of the first automated pattern recognition systems was developed by Martin in 1972. It employed a low-pass filtered for EEG at 28 Hz and EOG at 14 Hz [136]. The FFT analysis was performed on 30-second epochs to separate S2, S1 and W. Conjugate REMs were identified on 2-channel EOG. Pattern recognition was employed for the EEG delta band detection. Peaks ("the delta measurement program examined successive 30-second increments of EEG data to locate peaks no closer" [136]) were sensed as local maxima with no higher local maxima at 0.5 second. Valleys were detected as the lowest points between two adjacent peaks and peak-to-valley variation had to surpass 75  $\mu$ V, and the coefficient of correlation of the raw data in the fitted line (straight) from peak-to-valley was at least 0.75 [136]. An agreement of 82% was obtained in separating REM, S1, W, S2, S3 and S4, which was only 7% less than the inter-scorer agreement. Other recent automated methods including a semi-automated system as Morpheus [155] and automated Somnolyzer 24 X 7 [149] used in the clinical settings. The automated and semi-automated systems of sleep scoring provided an option for time-consuming, costly and variable manual scoring. According to scholars, the automatic analysis of SWS can be more reproducible than the visual analysis [156].

Van et al. [157] employed EEG automatic detection for sleep staging. The method comprised four steps, including segmentation, extraction of parameters, analysis of cluster (aimed to classify (cluster) a number of points into K groups), and classification. The parameters compared included the harmonic, Hjorth and relative band energy parameters. For cluster analysis, the study used a modified version of K-means algorithm. The study concluded that it was capable of extracting information from EEG pertinent for sleep stage scoring, and it was also possible to uncover similar segments and, therefore, automate the sleep detection stages through K-means algorithm. However, it was reported that extra information, such as EOG and ECG was critical for vibrant fine-tuning of various sleep stages [157]. Zhang et al. [158] undertook a study on automatic sleep onset detection utilising EEG electrodes. They proposed a simple and accurate method for sleep onset prediction that was dependent on the EEG signal, obtained from a single frontal electrode within a wireless headband. This method first extracted an energy power ratio of EEG alpha (8-12 Hz) and theta (4-8 Hz) bands along the time domain. The resultant slow waves were used for sleep onset detection by feeding them into a rule-based engine. The polysomnographic method was used to evaluate the effectiveness of the approach and also headband EEG signals were acquired from 20 healthy adult subjects who each underwent two sessions of sleep events; hence, 40 sleep events were collected. The resultant recordings were assessed by an offline PSG technologist through visual observation of the waveforms of PSGs

who annotated sleep stages N2 and N1 by using the AASM scoring rules. This approach achieved 87.5% accuracy when the gold standard was used in sleep onset detection. These results were better and comparable to other techniques employing single or multi-channel data [158].

Several other approaches have been reported based on single EEG channel data. Huang and colleagues [159] developed a system that detected the arousal states of individuals utilising mean frequencies of a single EEG for autoregressive hidden Markov models (HMM). This model accomplished wake-drowsiness detection at a rate of 70%. Novak et al. [160] used more features than the earlier system of Huang et al. [159], including autoregressive parameters, spectral entropy and the complexity stochastic measure in developing an HMM model used for sleep staging. This approach predicted sleep N4 and N3, but it was not able to detect well for stage REM. Rossow, et al. [161] also depicted an approach applying a single-channel EEG modelling using the HMM and Kalman filters. Its agreement rate when tested was 60.14%. Other researchers have adapted different approaches in the automatic determination of patients' sleep stages. Sukhorukova et al. [162] undertook a study to determine the difficulties and solutions of automatic sleep stage identification. They documented various problems, such as the data mining algorithms being challenging since the data set was noisy and colossal, and the signals were complex, thus needing specialists to analyse them. Therefore, the study adapted approaches from four fields involving mathematical optimisation, neural networks, frequency domain and financial forecasting to solve the problem identified in relation to the automatic determination of the patients' stage of sleep. The outcomes of this study, though preliminary, were promising and indicated that a combination of approaches can be more fruitful than relying on a single approach.

Research undertaken using 9-stage sleep onset automatic classifications showed that most hypnagogic imageries were remembered during the stage 5 EEG (theta stage) [163]. These imageries were least recurrent in stage 2 and 1 with EEG alpha activities. When the nine stages were classified depending on the subjective analysis of reaction time, recall rate and behavioral state of hypnagogic, the subgroups were not entirely coincidental. This denotes that sleep onset is influenced by the technique of determination. As the quantity of the subjective responses for having been asleep was 43.7 only in conventional S2, the sleep onset period can be considered as an extension beyond S1. "If the criterion for wakefulness is cognitive response to external stimulation, only in EEG Stages 3, 4, and REM can accurate distinctions between sleep and wakefulness be made. If EEG is the criterion, then the data suggest that cognitive response is possible during Stages 1 and 2 sleep" [163]. Therefore, it seems that the absence or presence of the response is an uncertain measure for sleep onset. Hence, the slowing of reaction time should be applied as an additional parameter [164]. Table 7 shows different automatic sleep stages algorithms, and some of these studies were described in detail [130].

Author(s)	Features	Classifier	Number of	Accuracy	Signals
			Subjects	(%)	_
Principe	frequency	Fuzzy logic	5 healthy	84.74	1-EEG,2-
(1989) [165]	bands, spindle,				EOG
	K-complex				EMG
Schwaibold	periodic	ANN	8 OSA patients	31.5 - 87.4	2-EEG
(2001) [149]	activity,				2-EOG
	spindles, K-				1-EMG
	complexes				
McGrogan	reflection	ANN	9 healthy, 3 non-	72.2 - 96	1 EEG
(2001) [166]	coefficients		healthy		
Flexer (2002)	reflection	HMM	40 healthy	68 - 82	1 EEG
[167]	coefficients,				
	stochastic				
	complexity				
Hanaoka	zero-crossing,	decision tree	1 healthy	40 - 91.8	1-EEG
(2002)	wave period,	learning			1-EOG
[168]	amplitude,	_			1-EMG

Table 7: An overview of automatic sleep staging methods and studies based on feature extraction, classifier, number of subjects, accuracy and signals.

-	1			-	-
	duration,				
11 (2002)	transient events		<b>C</b> 1 1/1	00.02	2 550
Huang $(2003)$	Lempel-Ziv	ANN	6 healthy	90.83	2-EEG
[109]	complexity				
Huunonen	amplitude	rule based	15 healthy	76.80	7 FEG
(2003)	spectrum of	Tule-Dased	15 licatury	70.00	7-LEO
[170]	subbands				
Gudmundsson	Hiorth	SVM. ANN	4 healthy	81	1 EEG
(2005)	parameters,	, , , , , , , , , , , , , , , , , , , ,	j	-	_
[171]	power				
	spectrum,				
	amplitude and				
	frequency				
	distribution				
Song (2007)	wavelet	rule- based	4 healthy	55.4 - 97.7	1 EEG
[172]	transform				
	modulus				
Virkkala et al	spectrum of	Decision	256 healthy and	74	2-EOG
(2008)	bandwidth	tree	non-healthy	/+	2-200
[137]	frequency	ucc	non neuriny		
[107]	distribution				
Güneş et al,	Welch spectral	K-NN &	5 healthy	82.15	1-EEG
(2010)	analysis	decision tree	2		
[173]					
Dong et al.	empirical mode	FFT	15 non healthy	80	1-EEG
(2010)	decomposition				
[175]	(EMD)				
	algorithm,				
	Fourier				
	non stationary				
	signals				
	Harmonic				
	components.				
Hedner et al.	PAT recorder	Fuzzy logic	227 including	59 - 94	2-EEG, 2-
(2011)	signals, Itamar	, ,	healthy & non-		EOG,
[174]	Medical,		healthy		EMG, ECG
	Caesarea				
Liang et al.	multiscale	LDA	20 healthy	76.91	1-EEG
(2012)	entropy (MSE)				
[176]	and				
	(AP) models				
Tagliazucchi	neriodic-	Binary	73 healthy	80	1-FFG-
et al (2012)	activity	support	75 licatiny	00	fMRI
[177]	spectrum of	vector			inter
	bandwidth,	machine,			
	fMRI scanner	multiclass			
Diego (2013)	spectral analysis	Fuzzy logic	33 non-healthy	84	2-EEG, 2-
[178]					EOG, EMG
Correa et al.	spectral	ANN	16 non-healthy	83.6 - 87.4	3-EEG
(2014)	analysis,Time				
[179]	analysis,				
	Wavelet				
Popovio at al	spectral analysis	Decision	20 healthy	80	1 FEG
(2014)	spectral analysis	tree	27 nearing	80	1-LEU
[180]					

### 3 : Subjects and data sets

#### 3.1 Subjects

This section describes the three PSG databases used in this research. The following section 3.1.1 describes the PSG napping or daytime PSG data. Sections 3.1.2 and 3.1.3 describe the whole-night PSG data.

#### 3.1.1 Data set No.1

This existing PSG napping data set consisted of 10 healthy adult male subjects ranging in age from 21 to 43 years (M = 29) (see Table 8). None of the subjects had a previously diagnosed sleep disorder. RMIT Ethics approval was granted for the recording of this PSG data and all subjects signed the consent letters. A single continuous PSG recording of 20 minutes in duration was undertaken for each subject. This existing data was provided by Dr. Dean Cvetkovic, which originated from ARC linkage grant (2005-2009). All six channels were utilised for automated sleep scoring undertaken by Hypnolab commercial software (SWS Soft, Italy, 2006-2008), but only 1 EEG channel (C3-A2) data was used in this analysis. All subjects were exposed to audio and photic stimulation under biofeedback operant conditioning, influencing wake and sleep states. Therefore, the influence of these stimuli may have caused the PSG transients to differ considerably as compared to 'normal' wake and sleep conditions. Also, this PSG data can be considered to be recorded from 20 minute napping conditions rather than 20 minute sleep/wake conditions. Sleep stage 1 (N1) and wake (W) were detected during these napping conditions. The PSG recording for each subject contained 6 channels, 2 EEGs (O2-A1 and C3-A2), 2 EOGs (right outer canthus (ROC) and left outer canthus (LOC)) and an ECG (Lead II using torso electrode placement). The sample frequency was 256 Hz for each signal.

Table 8: Summary of the PSG napping data.

Type of subjects	Healthy
Number of subjects	10
Gender (male/female)	(10/0)
Age (mean years and range)	29 (21-43)
TRT (min)	195
W (% of TRT <sup>1</sup> )	92.1
N1 (% of TRT)	7.9

 ${}^{1}TRT = \text{total recording time}$ 

#### 3.1.2 Data set No.2

The existing PSG data was downloaded from the online database [181]. This data is under the terms and conditions of the Attribution-NonCommercial-NoDerivs 3.0 Unported (CC BY-NC-ND 3.0) License. Each use of one of these databases and/or its content is attributed to the University of MONS - TCTS Laboratory (Stéphanie Devuyst, Thierry Dutoit) and Université Libre de Bruxelles - CHU de Charleroi Sleep Laboratory (Myriam Kerkhofs). Each PSG data file consists of at least 2 EOG, 3 EEG and 1 EMG submental channel, at the sampling frequency of 200 Hz. All sensitive information was coded by this institute and the data does not show any patient names, except their group, age and gender. All PSG data files are up to 8-9 hours of recording of overnight sleep. The overnight PSG data was recorded from 13 healthy adult subjects, 10 with periodic leg movement syndrome (PLMS), 10 with sleep apnoeahypopnoea syndrome (SAHS) as shown in Table 9. The sleep stages were manually scored by an expert according to the AASM criteria.

Group of subjects	Group1: Healthy control	Group2: PLMS	Group3: SAHS
Number of subjects	13	10	10
Gender (male/female)	(3/10)	(8/2)	(6/4)
Age (mean years and range)	42.5 (20-65)	49 (22-46)	56 (38-74)
TRT (Min)	6627.8	5262.8	5249
W (% of TRT <sup>1</sup> )	16.8	25.2	25.4
N1 (% of TRT)	6.6	8.8	8.8
N2 (% of TRT)	45.7	47.5	51.5
N3 (% of TRT)	16.3	7.8	5
R (% of TRT)	14.6	10.7	9.3

Table 9: Summary of PSG data set 2.

 $^{1}$ TRT = total recording time

#### 3.1.3 Data set No.3

The PSG dataset was recorded by the Central Institute of Mental Health (Mannheim, Germany). Each donated PSG patient data file consisted of up to 15 channels: 3 EEG, ECG, EMG, 2 EOG, nasal and oral airflow, snoring sound, breathing effort (measured at the chest and abdomen), oximetry and actigraphy recording body positioning and leg movements. The sampling frequency varied from 200, 256 to 500 Hz. In addition to each European data format (EDF) file, the institute's sleep technicians manually/visually scored the sleep stages according to the AASM criteria.

The PSG database included 30 subjects who were divided into three groups, each consisting of 5 males and 5 females (see Table 10). Group 1 consisted of 10 primary insomnia patients, in Group 2 was 10 schizophrenia patients and Group 3 contained 10 healthy subjects. All patients and subjects were diagnosed on the basis of the Diagnostic and Statistical Manual (DSM-IV) and were recruited during in-house treatment at the Central Institute for Mental Health. The criteria for inclusion in the study included being aged between 18 and 60 years, the ability to provide informed consent, a stabilised disease course, stable (for at least two weeks) psychopharmacological treatment in the form of monotherapy with a second generation antipsychotic, and the absence of psychiatric comorbidity.

#### Table 10: Summary of PSG data set 3.

Type of subjects	Group 1: Primary insomnia	Group 2: Schizophrenia	Group 3: Healthy
Number of subjects	10	10	10
Gender (male/female)	(5/5)	(5/5)	(5/5)
Age (mean age and	31.3 (18-45)	31.7 (22-46)	31 (20-47)
range)			
TRT (Min)	4647	4899.5	4870
W (% of $TRT^1$ )	22.1	14.05	22.98
N1 (% of TRT)	10.2	7.95	9.34
N2 (% of TRT)	40.6	43.06	42.45
N3 (% of TRT)	18.2	18.28	12.77
R (% of TRT)	8.9	16.63	12.75

 $^{1}$ TRT = total recording time

## 4 : Automatic sleep stage detection

### methods

This chapter describes various methods and materials utilised to develop an automatic detection of sleep stages. Section 4.1 describes the materials and methods utilised to develop an automatic detection of wake and sleep stage N1 based on EEG sub-epoch signal. Section 4.2 describes the materials and methods that were utilised to develop an automatic detection of sleep stage based on the consecutive and non-consecutive EEG sub-epoch approach. Section 4.3 describes the materials and methods utilised to develop an automatic detection of sleep stages based on EOG signals.

# 4.1 Automatic detection of wake and sleep stage N1 using EEG sub-epoch approach

#### 4.1.1 Subjects

Ten healthy adult male subjects were used, as described in Section 3.1.1

#### 4.1.2 EEG signal processing method

Figure 8 shows the flowchart for the proposed consecutive and non-consecutive, 6second sub-epoch comparison approach comprised of six parts: EEG segmentation of the 30-second epochs (part 1); filtering (part 2); EEG segmentation of the 30-second epochs into 6-second sub-epochs (part 3); feature extraction (part 4); sleep stage scoring rules (part 5); and checking three consecutive and non-consecutive 6-second subepochs (part 6).

Part 1 segmentation of the 30-second epochs: The full length of this data was segmented into 30-second epochs for scoring the sleep EEGs prior to pre-processing.

Part 2 Filtering: The EEG signal was processed utilising a 6-order Butterworth band pass filter for the different band frequencies, as follows: theta ( $\theta$ , 4–7Hz), alpha ( $\alpha$ , 8–12Hz) and beta ( $\beta$ , 13–30Hz).

Part 3 Further segmentation of each of these 30-second epochs into 6-second subepochs: Prior to feature extraction, the 30-second epochs were divided into five subepochs of six seconds each.



Figure 8: Flow-chart describing the consecutive and non-consecutive, 6-second EEG epoch approach.

Part 4 Feature extraction: Frequency domain feature extraction was applied, calculating the relative spectral energy band (RSEB) for the  $\theta$ ,  $\alpha$  and  $\beta$  EEG bands from the were computed of parametric and non-parametric power spectral densities

 $(PSD_{\theta}, PSD_{\alpha}, PSD_{\beta})$ . The RSEB can be calculated as follows:

$$Total power = PSD_{\theta} + PSD_{\alpha} + PSD_{\beta}$$
(1)

 $RSEB(\theta) = PSD_{\theta} / Total power$ (2)

$$RSEB(\alpha) = PSD_{\alpha} / \text{Total power}$$
(3)

Part 4.1 Non-parametric method: Thomson's multi-taper method (MTM) [182] was used to estimate the PSD of each EEG band. The advantage of using the MTM is that it reduces the variance of the spectral estimate by utilising a few groups of tapers, as proposed by Thomson (1982) and Percival and Walden (1992) [183]. The computation of the power multi-taper method (PMTM) was put forward in [184].

Part 4.2 Parametric method: Non-parametric methods have the drawback of spectral leakage effects because of windowing, which can result in a weakening of the signal components. Using parametric methods to compute PSD is a solution to spectral leakage and provides optimal frequency resolution. In this study, Burg's autoregressive (AR) method was applied in this analysis to calculate the PSD. The main concept of Burg's method is that it attempts to reduce the forward and backward prediction errors by accepting the Levinson-Durbin recursion [185]. The reflection coefficients are estimated directly in Burg's method rather than calculating the autocorrelation function.

Part 5 Sleep stage scoring rules: In this part, rules for scoring the sleep stages in each 6-second epoch were applied. For example, according to the R&K rule, an epoch is scored as a W stage if RSEB( $\alpha$ ) represents more than 50% of the 30-second epoch. On the other hand, the epoch is scored as stage 1 when the RSEB( $\theta$ ) is more than 50% of the epoch. In addition, N1 can be scored when the RSEB( $\alpha$ ) is less than 50% of the epoch.

Part 6 Checking the three consecutive and non-consecutive 6-second sub-epoch: This was accomplished by investigating whether three out of five consecutive or non-consecutive 6-second sub-epochs indicated W or N1 sleep. The three out of five consecutive and non-consecutive 6-second sub-epochs characterised 60% of

predominant band power in its total, 30-second epoch, making the detection more difficult based on the rule criterion of 50% predominant power.

#### 4.2 Automatic detection of sleep stages using EEG sub-epoch approach

#### 4.2.1 Subjects

Thirty-nine subjects were used in this study. Ten healthy subjects were utilised for the development of this algorithm (see section 3.1.2) and 29 subjects were utilised for the validation (see section 3.1.3)

#### 4.2.2 Automatic sleep stages detection algorithm

Algorithms for automated sleep stage detection were previously explored with varying degrees of success (the algorithm was succeed with healthy subjects of varying age) [186]. Here, the candidature focused on the approach outlined in [186], using a single EEG (CZ-A1) channel and validated against the AASM criteria. An algorithm, outlined in Figure 9, involves a consecutive or non-consecutive sub-epoch approach, and seven steps that are summarised as: (1) preprocessing; (2) EEG signal segmentation into 30-second epochs; (3) 30-second epoch split into five, 6-second long sub-epochs; (4) filtering; (5) feature extraction within sub-epochs; (6) using three consecutive or non-consecutive 6-second sub-epochs; and (7) a 'smoothing' rule. The algorithm introduces several new aspects including the new segmentation strategy and application of an adaptive filter to remove electrocardiogram (ECG) artifacts from the EEG signal. The following subsections detail the explanation of the specific procedures:

#### 1. Preprocessing and EEG Signal Segmentation

During initial preprocessing, a baseline drift is removed, eliminating any linear trends and removing the mean value. As mentioned in section 3.1, the EEG data was segmented into 30-second epochs. Additionally, since EEG signals can be

affected by ECG, an adaptive filter based on a least mean square (LMS) algorithm was employed to eliminate artifacts from the EEG data. Subsequently, the 30-sec EEG signal epochs were divided further into five sub-epochs of 6-second duration each,



Figure 9: Flow chart describing the automated sleep stage detection algorithm based on the three consecutive and non-consecutive 6-second EEG sub-epochs approach. Where the '\*' is indicated to 3 consecutive or non-consecutive out 5 sub-epoch approach.

#### 2. Filtering

A zero-phase bandpass filter was applied to the EEG data. The filter was a sixthorder cascaded integrator-comb (CIC) type that selects the following EEG frequency bands of interest: delta 1 ( $\delta$ 1, 0.5–2 Hz), delta 2 ( $\delta$ 2, 2–4 Hz), saw-tooth
waves (2-6Hz), theta ( $\theta$ , 4–8 Hz), alpha ( $\alpha$ , 8–13 Hz), sigma ( $\sigma$ , 12–15 Hz), beta 1 ( $\beta$ 1, 13–20 Hz) and beta 2 ( $\beta$ 2, 20–30 Hz).

#### 3. EEG Feature Extraction

In order to extract features from the 6-second long EEG signal sub-epochs, spectral analysis was used. Burg's parametric method was used to calculate the power spectral density (PSD) and estimate the spectral content by an autoregressive linear prediction filter model. Suitability of this approach was confirmed by the Akaike information criterion (a test effective in assessing the relatively quality of models) [187]. The parametric method was chosen because of its ability to resolve spectral leakage caused by 'windowing' and because it provides optimal frequencies.

**Relative spectral energy (RSE)** of the *i*th frequency band,  $B_i$ , is the sum of power within the band's frequency range relative to the overall power in the entire considered spectrum, which can be written as:

$$RSE_{i} = \frac{\sum_{f \in B_{i}} P(f)}{\sum_{f_{L}}^{f_{H}} P(f)}$$
(5)

where  $f_L$  represents the lowest (0.5Hz) frequency and  $f_H$  represents the highest frequency (30 Hz) contained in the EEG signal. P(f) is the power at a given frequency, f.

The power spectral density ratios (PSDR) is defined, as (an example):

$$\frac{PSDR_{\alpha}}{PSDR_{\delta}} = \frac{\sum_{f=8Hz}^{12Hz} P(f)}{\sum_{f=0.5Hz}^{2Hz} P(f)}$$
(6)

#### Central frequency (CF) is determined by

$$FC = \frac{\sum_{f_L}^{f_H} f * P(f)}{\sum_{f_L}^{f_H} P(f)}$$
(7)

#### The 90% Spectral Edge Frequency (SEF) is

$$SEF = 0.9 \sum_{f_L}^{f_H} P(f)$$
 (8)

where L is the 'length' of the RSE array and is determined by a number of frequency 'windows' used in the calculation.

Root mean square frequency (RMSF) was defined as:

$$RMSF = \sqrt{\frac{\begin{array}{c} f_{H} \\ \Sigma \\ f_{L} \end{array}}{\begin{array}{c} f_{H} \\ f_{L} \end{array}}} (9)$$

#### 4. The Three Consecutive and Non-Consecutive 6-Second Sub-Epochs Approach

Varying segmentation strategy can have a drastic effect on the efficacy of algorithms. Segment length is not the only important factor. Here, the purpose of this step is to investigate whether three (out of five) consecutive or non-consecutive 6-second sub-epochs are most effective at detecting the sleep stages. Since three out of five consecutive (or non-consecutive) 6-second sub-epochs contain only 60% of the predominant band power in the total 30-second epoch, detection could be expected to be more difficult, based on the rule of 50% predominant power [185]. However, the approach can be surprisingly effective, when the appropriate technique is applied.

#### Identification of Sleep Stage Based on the Three Consecutive and Non-Consecutive 6-Second Sub-Epochs Approach

Increasing the number of features used to identify each sleep stage is known to offer better detection performance [153,188,189]. Table 11 shows the sleep stage features and the associated rules used in this study. The algorithm process is outlined in Figure 9.

At first, each rule was utilised for each sleep stage and both the three consecutive and non-consecutive 6-second sub-epoch method. Initially, RSE was applied because these features are well known from the AASM and R&K rules for the detection of sleep stages. For example, when RSE ( $\alpha$ ) represents more than 50% of the epoch's total spectral energy, then rule 1 (R1) is positive and the epoch is scored as the W (waking) stage. For example, in the first step (R1) RSE ( $\alpha$ ) feature was used to split the W from the other sleep stages (see Figure 2). In the second step the slow-wave sleep (N3) was split from sleep stages, N1, N2 and R, using rule 4 (R4) and RSE ( $\delta$ 1), as this sleep stage is characterised by lower EEG delta activity. In the third step, sleep stage N2 is split off from sleep stage N1 and R by rule 3 (R3) as this sleep stage is characterised by sleep spindles. Sleep stage R is characterised by saw-tooth wave and was split from N1 in the fourth step by rule 5 (R5). A similar procedure was done for the power ratio features in the event of negative rule outcomes in the previous steps. The remaining

rules (R11, R14-R17) and associated features represent contrasts between neighbouring stages and may be needed prior the sub-epoch being identified as a particular sleep stage.

Table 1	1: Sleep stage a	lgorithm ru	les.

		W	N1	N2	N3	R
	AASM rule [13]	EEG Alpha >50 % of epoch. If alpha rhythm is present for part of the epoch (even <15 sec) score as wake stage	EEG Theta >50% of epoch or EEG Alpha <50 % of epoch	Present sleep spindle & k-complex	EEG Delta >20 % of epoch	Low amplitude mixed frequency EEG Theta, Low chin EMG Rapid eye movement EOG
lle	RSE feature [190]	<b>R1</b> :RSE(α) >50% of epoch	<b>R2</b> :RSE (δ2) <15% and RSE (β2)>10% of epoch or RSE(θ)> 50%	<b>R3</b> : RSE $(\delta 2) <20\%$ & RSE $(\beta 2) <10\%$ of epoch, or RSE $(\sigma) >10\%$ . respectively	<b>R4</b> : RSE (δ1) >20% or >50 of epoch	<b>R5</b> : RSE (saw-tooth wave) >.5% Or RSE ( $\beta$ ) and ( $\sigma$ ) <10 % of epoch, RSE ( $\beta$ 2) approximately 10% and RSE ( $\delta$ 2) <15 %
hm Ru	Power ratio	<mark>R6</mark> : α /β1> 7.8	<mark>R7</mark> : α /β1<2.5	<mark>R8</mark> : σ/ β2 >3.5	<b>R9</b> : δ1/θ>5.5	$\frac{\mathbf{R10}}{\delta 1/\theta} < 1.6$
Algorit	Central frequency	$\frac{\text{R11:}}{\text{CF} (\alpha) > 40\% \text{ of epoch}}$	-	-	-	-
	Root mean square spectrum	R12: RMSF (α ) > 40 % of epoch	<b>R13</b> : RMSF (α) >19 % & < 40 %	-	-	-
	Spectral edge frequency (90 %)	R14: SEF >=8 and <=13 Or SEF 20> and <=30	<b>R15</b> : SEF >=4 and <=8	R16: SEF >=12 and <=15	<b>R17</b> : SEF>=.5 and <=2	-



Figure 10: Flow chart depicting process for sleep stage identification applied for both the three consective and non-consecutive epoch approaches.

#### 6. The Smoothing Rule

Due to fluctuations in the recorded signals, it is desirable to 'smooth' the outputted sleep stage values to reduce output 'jitter' (misclassification of sleep stage, see Table 12 for the list of smoothing rules). For example, if the case detection of the sleep stage was a consecutive sequence such as W- (any stage)-W then, the consecutive sleep stage value should be replaced as W-W (e.g. see rule number 1, Table 12).

Table 12: The smoothing rules

Rule number	Case detection	Replace to consecutive sleep stage
1	W-(any stage)-W	W-W-W
2	N1-(any stage)-N1	N1-N1-N1
3	N2-(any stage)-N2	N2-N2-N2
4	N3-(any stage)-N3	N3-N3-N3
5	R-(any stage)-R	R-R-R

#### 4.3 Automatic detection of sleep stages using EOG

#### 4.3.1 Subjects

Thirty subjects and patients were used in this study, as described in section 3.1.

#### 4.3.2 EOG signal Method

Figure 11 describes the algorithm for detecting the sleep stages based on EOG signals.

It includes five steps, as shown in Figure 11.



Figure 11: Block diagram of the automatic sleep stage detection method using EOG signals.

#### 4.3.2.1 Pre-processing

The EOG data was segmented into 5-second epochs. The entire EOG signal was processed utilising a zero-phase bandpass filter with a cascaded integrator-comb (CIC) filter of order 6, for the following different frequency bands: delta (0.5-2 Hz), delta1 (2-4 Hz), theta (4-8 Hz), alpha (8-12 Hz), sigma (12-16 Hz), beta1 (16-20 Hz), and beta2 (20-30 Hz). Since an EOG signal might be affected by EMG, and ECG artifacts, launching a suitable algorithm in order to remove the artifacts and noise was necessary.

A cascade of three adaptive filters based on a least mean square (LMS) algorithm was employed, to eliminate ECG, and EMG artifacts [191].

#### 4.3.2.2 Feature extractions

Several features were extracted from the EOG signal in the time and frequency domain, such as variance, the maximal peak amplitude value (MAX-PAV), the minimum peak amplitude value (MIM-PAV), energy entropy, Shannon entropy, and cross-correlation. In order to select the best feature that classified the variations in sleep stages and wakefulness, the sequential feature selection method (SFS) was used.

The definitions of the features are as follows :

#### Energy entropy (EE)

The energy entropy was performed, utilising wavelet packets as

$$EE(s) = \sum_{i} \log(s_i^2) \tag{10}$$

Where (s) is the EOG signals and  $s_i$  is the coefficient of (s).

#### Shannon entropy (SE)

The Shannon entropy was performed by utilising wavelet packets as

$$SE(s) = -\sum_{i} s_{i}^{2} \log(s_{i}^{2})$$
(11)

Where (*s*) is the EOG signals and  $s_i$  is the coefficient of (s).

#### Cross correlation

For each 5-second epoch, the cross-correlation between each frequency band of EOG left, EOG right channels, and peak-to- peak amplitude differences from EOG left – right were calculated [137].

#### Maximum peak amplitude value

The maximum peak amplitude value (MAX-PAV) is a measure of the EOG signal amplitude value at the highest point (i.e. the maximum positive value) in both EOG right and EOG left signals.

#### Minimum peak amplitude value

The minimum peak amplitude value (MIM-PAV) is a measure of the EOG signal amplitude value at the lowest point, (i.e the minimum negative value), EOG right and EOG left.

#### Variance

The variance of the EOG signal is a measure of the signal power calculated as

$$VAR = \frac{1}{N} \sum_{i=1}^{N} (x_i - \bar{x})^2$$

Where  $x_i$  is ith sample of EOG signal,  $\overline{x}$  is the mean and 'N' is the epoch size for computing features.

#### 4.3.2.3 Classification using K-nearest neighbour classifier

In this study, the K-nearest neighbour (KNN) classifier was used for classification of the sleep and wake stages. The KNN is based on a nonparametric method and can be employed for a different pattern classification approach, which represents as one robust classifier [192]. The KNN classifier is based on a comparison between a new sample (testing data) and baseline (training data). It attempts to find out the K-nearest neighbour within the baseline, and indicates a class which seems more normally in the nearest neighbour of K. The value of K might need to be diverse in order to detect the corresponding class between the training and testing data. In this study, the value of K varies from 1 to 5. The Euclidean distance metric is utilised for calculating the distance between the two points. The training and testing data was evaluated based on 10-fold cross-validation.

#### 4.3.2.4 Smoothing rule

The smoothing rule is used to increase the detection accuracy of the sleep stages (for more information see Table 12).

#### 4.4 Sleep architecture statistical method

In order to evaluate the performance of the algorithms for automatic sleep stage detection, the sensitivity and specificity were calculated for each sleep stage. The interrater agreement (IRA) and Cohen's Kappa were calculated in order to evaluate the overall performance of the algorithms. In this dissertation the criteria of calculation sensitivity and specificity methods that were utilised for the sleep stages was done according to Devuys et al. criteria [193].

$$\blacktriangleright \quad Sensitivity = \frac{TP}{TP + FN} \tag{12}$$

$$\succ Specificity = \frac{TN}{TN + FP}$$
(13)

where TP is true positive; FN is false negative; FP is false positive; TN is true negative.

The IRA was calculated as the following:

$$IRA = \frac{The \ total \ correct \ stage \ detection}{Total \ stages} \tag{14}$$

The Cohen's Kappa was utilised to evaluate the IRA matrix, 'k', which can be calculated as [194]:

• 
$$K = \frac{p_o - p_c}{N - p_c}$$
(15)

where  $p_o$  is the total correct stage detection (sum of the diagonal), 'N' total stages and  $p_c$  is the expected frequency for the number of agreements that would be expected by chance for each stage.

The interpretation of Cohen's Kappa coefficient was as the follow: < 0.00 (poor agreement); 0.00 - 0.20 (slight agreement); 0.21- 0.40 (fair agreement); 0.41- 0.60 (moderate agreement); 0.61- 0.80 (substantial agreement); > 0.80 (excellent agreement) [194].

# 5 : Automatic classification of sleep

### disorders

#### 5.1 Classification of primary insomnia, schizophrenia and healthy sleep

#### 5.1.1 Subjects

Thirty subjects participated in this study (see date set No. 3).

#### 5.1.2 Method

5.1.2.1 Calculating PSG sleep stage parameters

In this study, we used the visual sleep stage scoring according to the AASM rules to measure different PSG sleep stage parameters, such as sleep latency (SL), sleep efficiency (SE), number of awakenings (NW), total sleep time (TST), waking after sleep onset (WASO), slow wave sleep (SWS), REM sleep, the first REM period, and the characteristics of the sleep stages (N1, N2, N3 and R).

#### 5.1.2.2 Decision tree analysis

A decision tree analysis was used to classify the three groups based on the following rules:

- R1 was used for the first REM period parameter to separate primary insomnia from schizophrenia patients. If R1 (first REM period) was less than two minutes then it would be classified as primary insomnia; if R1 (first REM period) was more than two minutes and less than 10 minutes it was classified as schizophrenia.
- R2 was used for the N1 sleep stage parameter to identify the primary insomnia patients that R1 was unable to identify. If R1 (first REM period) was more than five minutes and R2 (sleep stage 1) was more than 12.5 % of the total time of sleep then it was classified as primary insomnia.
- R3 was utilised for SL to distinguish between the schizophrenia and healthy (control) subjects. If R3 (SL) was less than 19 minutes, then it was classified as a

healthy control. All of these rules were based on some percentage of sleep stage and time duration of sleep parameters, such as SL and the first REM period. The automatic classification algorithm is described in Figure 12.



Figure 12: Flow chart describing the automatic classification algorithm for primary insomnia, schizophrenia and healthy (control) patients based on the decision tree analysis. R1 is characterised by the first REM period in minutes, R2 by the % of sleep stage 1, and R3 by the SL in minutes.

#### 5.1.2.3 Statistical analysis

A statistical analysis was conducted using post-hoc tests to ascertain whether there were significant differences between the three groups. Sensitivity and specificity tests as well as Cohen's Kappa were conducted to evaluate the automatic classification algorithm for the three groups.

#### 5.2 Classification of PLMS, SAHS and healthy sleep

#### 5.2.1 Subjects

Thirty subjects were utilised in this study (see data set No. 2 in section 3.1.2).

#### 5.2.2 Method

5.2.2.1 Computing PSG sleep stage parameters

In order to classify the PLSM, SAHS, and healthy (control) subjects, different PSG sleep stage parameters were first computed, such as SL, SE, the number of times the subject woke up (NW), TST, WASO, SWS, and REM. During the first REM period, characteristics of the sleep stages (N1, N2, N3 and R) were based on the automatic detection system detailed in the previous section (4.3).

5.2.2.2 Decision tree analysis

A decision tree analysis was preformed to classify the three groups of subjects based on the following rules:

- Rule (1) used the percentage of the sleep stage N1 parameter to separate the SAHS patients from the PLMS patients and healthy (control) subjects. If N1 (%) was more than 7% and less than 9% of the total sleep time, then a subject was classified as an SHAS patient; if N1 (%) was less than 4% of the total sleep time, then a subject was classified as a healthy (control) subject; if N1 (%) was more than 10% of the total sleep time, then a subject was classified as a healthy (control) subject; if N1 (%) was more than 10% of the total sleep time, then a subject was classified as a PLMS patient.
- Rule (2) used the percentage of the N2 sleep stage parameter to separate the SHAS from PLMS patients. If N2 (%) was more than 80% of the total sleep, a patient was classified as an SAHS patient; If N2 (%) was less than 80% and more than 60%, of the total sleep, a patient was classified as a PLMS patient.
- Rule (3) utilised slow wave sleep duration (SWSD) in minutes and the percentage of the N3 sleep stage to discriminate between PLMS and healthy (control) subjects. If SWSD was more than 70 minutes and N3 (%) was more than 20% of the total sleep, a subject was classified as a healthy (control) subject. All of these rules were based on a percentage of the sleep stage and the time duration of the sleep

parameters, such as N1%, N2%, and SWSD. The automatic classification algorithm is described in Figure 13.



Figure 13: Flow chart based on decision tree analysis describing the automatic classification algorithm for PLMS and SAHS patients, and healthy subjects. N1 and N2 are the sleep stage percentages and SWSD is the duration in minutes.

#### 5.2.2.3 Statistical analysis

The statistical method was mentioned in section 5.1.2.3.

### 6 : Results

#### 6.1 Automatic detection of wake and sleep stage N1 using EEG sub-epoch approach

The aim of this study was to detect the W and sleep S1 based on consecutive and nonconsecutive sub-epoch of EEG, and then compare the performance of the conventional 30-second epochs with the three consecutive and non-consecutive 6-second epochs for detection W and sleep S1. As an example Figure 14 shows the hypnograms for subject 22, using the parametric methods. The figure also shows the difference between using 30-second epochs and three consecutive and non-consecutive 6-second epochs for identification of the W and sleep S1. From Figure 14, it is clear that the three consecutive and non-consecutive 6-second sub-epochs provide slightly better results than the 30-second epoch method. In addition, the non-parametric method showed lower detection accuracy for the W stage but higher detection accuracy for sleep S1. It is clear from Table 13 that the non-consecutive 6-second epoch method.



Figure 14: The hypnograms for subject S22 using the parametric approach, comparing the 30-second and the three consecutive and non-consecutive 6-second epochs methods. 'W' = the wake stage; 'S1' = stage 1; 0 = misclassification.

Table 13 describes the count of W and stage S1 epochs automatically detected by the parametric and non-parametric methods, and compared to the commercial 'Hypnolab' tool. The 20-minute PSG recording translated into 39 epochs, each epoch consisting of 30-second epochs (the last 30 seconds were not scored). While the sum of the W and S1 epochs was expected in all subjects (there were no other stages present), the W stages were all accounted for both parametric and non parametric and its three methods (30-second epoch, consecutive and non-consecutive sub-epochs). However, for S1, not all epochs were detected with the three epoch/sub-epoch methods. Some subjects did not have any stages detected by Hypnolab, such as S18, S23, S29 and S30. Therefore, the validation for S1 for those subjects was not available (noted by N/A in Table 13). Also, for some subjects the consecutive and non-consecutive sub-epoch methods showed instances of not classified or misclassified, noted by M/C in Table 13.

		l	Non-Pai	ametri	ic				Paran	netric			'Hypnolab'		
	W	ake sta	ige ochs	SCO	Stage red en	1 Jochs	W	ake stag	ge chs	SCO	Stage	1 ochs	Wake	& Stage	
po	500	rea ept	Jens	r			500	·····						i scoled epoen	
Meth	30-s epoch	onsecutive 6- s sub-epoch	on-consecutive -s sub-epoch	30-s epoch	onsecutive 6- s sub-epoch	on-consecutive -s sub-epoch	30-s epoch	onsecutive 6- s sub-epoch	on-consecutive -s sub-epoch	30-s epoch	onsecutive 6- s sub-epoch	on-consecutive -s sub-epoch	Wake 30-s scored	Stage 1 30-s scored epochs	
Subject		3 с	3 n 6		3 с	3 n 6		3с	3 n 6		3 с	3 n 6			
S18	39	38	39	N/ A	N/ A	N/A	39	36	39	N/ A	N/ A	N/A	39	N/A	
S19	13	12	21	26	9	18	35	26	35	4	M/ C	4	34	5	
S20	27	14	20	12	11	19	39	23	35	M/ C	M/ C	4	33	6	
S21	25	30	36	14	1	3	39	30	37	M/ C	M/ C	2	35	4	
S22	35	25	36	4	2	3	39	30	37	M/ C	M/ C	2	35	4	
S23	8	2	6	31	2	33	39	31	37	M/ C	1	2	39	N/A	
S25	33	16	26	6	M/ C	13	39	30	38	M/ C	M/ C	1	38	1	
S27	34	34	38	5	1	1	39	28	36	M/ C	1	3	28	11	
S29	33	25	34	6	M/ C	5	39	33	39	M/ C	M/ C	-	39	N/A	

Table 13: The count of W and S1 stages scored epochs from the three automated detection methods using parametric and non-parametric spectral density functions, and compared with the commercial 'hypnolab' sleep scoring.

S30	33	36	39	6	M/	M/C	39	31	38	M/	M/	1	39	N/A
					С					С	С			
Mean	28	23	29	12	4.3	11.8	38.6	29.8	37.1	4	1	2.37	35.9	5.1
S.D. 10 11 10 9 4.4 11 1.26 3.58 1.44 0 0 1.18 3.6 3.3														
Note N/A	Nota: $N/A = not$ available: $M/C = not$ classified or misclassified													

*Note:* N/A = not available; M/C = not classified or misclassified.

Table 14 revealed achieved an accuracy with the non-parametric spectral density 30second epoch method of 71.4% in the wake stage and 47.5% in stage 1, while the accuracy of the three consecutive 6-second sub-epoch method was slightly lower than the 30-second method, as 59.3% in the wake stage and 35% in stage 1. The three nonconsecutive 6-second method revealed a slight improvement in accuracy as compared to the 30-second epoch method (non-parametric), at 77% (increase by 5.6%) for the wake stage and 55.8% (increase by 8.3%) for S1. Moreover, for some subjects the three consecutive and non-detection method for S1 proved to be more effective over the conventional 30 sec epoch method. For example, from Table 14, the accuracy of the detection of S1 for subjects S20 and S22 was higher by 80% with the three consecutive and non-consecutive 6-seconds methods. While, the accuracy of the 30-second method was 60% for subject S20 and 25% for subject S22, the average accuracy for all 10 subjects was higher with the 30-second method as compared to the consecutive and non-consecutive methods. This standard deviation was also large in these subjects with the SD at 25-30% for the 30-second epoch and 40-46% for the consecutive and nonconsecutive 6-second sub-epoch methods.

On the other hand, the parametric method revealed a significant accuracy in the detection of the W stage, wherein the accuracy of the 30-second method was 98.5%, the three consecutive 6-second method was 75.6% and the accuracy of the three non-consecutive 6-second method was 95.1%. The S1 detection was 0% for both the 30-second epoch and 6-second consecutive sub-epoch methods, and 10.8% for the 6-second non-consecutive sub-epoch.

Method			Non-Pa	arametri	c		Parametric						
	30-s e	epoch		3	3 no	on-	30-s	epoch		3	3 no	on-	
		1	conse	cutive	consecutive			1	conse	cutive	consecutive		
			6-s sub-		6-s s	sub-			6-s sub-		6-s s	sub-	
			epo	ochs	epo	chs			epo	ochs	epo	chs	
Subject	W	<b>S</b> 1	W	S1	W	<b>S</b> 1	W	<b>S</b> 1	W	<b>S</b> 1	W	S1	
/W&S1	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	(%)	
%(													
S18	100	N/A	97.4	N/A	100	N/A	100	N/A	92.3	N/A	100	N/A	
S19	38.2	100	32.3	80	58.8	80	88.2	0	67.6	0	88.2	0	
S20	70.5	60	35.2	80	53.9	80	97	0	61.7	0	91.1	40	
S21	60.0	0	74.2	0	91.4	0	100	0	74.2	0	94.2	0	
S22	91.4	25	68.8	50	100	75	100	0	77.1	0	97.1	25	
S23	20.5	N/A	5.1	N/A	15.3	N/A	100	N/A	79.4	-	97.4	-	
S25	86.8	100	42.1	0	68.4	100	100	0	76.3	0	97.3	0	
S27	82.1	0	82.1	0	96.4	0	100	0	64.2	0	89.2	0	
S29	84.6	N/A	64.1	N/A	87.1	N/A	100	N/A	84.6	N/A	100	N/A	
S30	84.6	N/A	92.3	N/A	100	N/A	100	N/A	79.4	N/A	97.4	N/A	
Mean	71.4	47.5	59.3	35	77	55.8	98.5	0	75.6	0	95.1	10.8	
S.D.	25.2	46.2	28.1	39.8	27.9	44.9	3.74	0	9.3	0	4.3	17.4	

Table 14: The comparison of the accuracy between the three automated detection methods using nonparametric and parametric spectral density functions.

*Note:* N/A = not available in subject file.

#### 6.2 Automatic detection of sleep stages using EEG sub-epoch approach

#### **6.2.1 Overall Performance**

To help gauge the overall effectiveness of this study's automatic detection approach, it was compared with the visual scoring method done by an expert (see Table 15 for the results of the performance and comparison). The rows and columns of the confusion matrix show the results of visual detection and automatic detection, respectively. Whilst the three consecutive sub-epoch approach yielded only 64.2% (0.50 Cohen's Kappa) epoch-by-epoch agreement detection of the 5-stages for group 1, the three non-consecutive sub-epoch method gave 82.4% (0.75 Cohen's Kappa). The situation was similar for Group 2, where the three consecutive and non-consecutive sub-epoch method agreement was 71% (Cohen's Kappa = 0.62) and 86.6% (Cohen's Kappa = 0.82), respectively. Likewise, in Group 3, the overall agreement of the three consecutive and non- consecutive sub-epoch method with visual detection was 68.35% (0.60 Cohen's Kappa) and 80.85% (0.74 Cohen's Kappa), respectively.

When comparing the visual detection hypnogram to the automatic sleep stages determined by the 30-second epochs (for patient no. 8 from Group 2), a relatively poor comparison can be seen as in Figure 15. On the other hand, the automated three consecutive and non-consecutive 6-second sub-epoch methods were closer to the hypnogram that was visually scored. Upon closer inspection, when considering the finer details between the 300 and 400 epoch numbers, it can be clearly seen that the three non-consecutive approaches were more closely matched with the visual detection approach than the other methods.

#### 6.2.2 Comparison of sleep stage detection method

As shown in Table 15, the sensitivity and specificity of the five stages for each group of patients or subjects was better for the three non-consecutive 6-second sub-epoch method than the three consecutive 6-second sub-epoch method. For example, for Group 1 the performance of the three non-consecutive method for all stages (stage (sensitivity, specificity)) was W (84.8, 89.9), N1 (68.83, 89.8), N2 (87.7, 96.7), N3 (73.4, 97.4) and R (83, 98.8), and similar to the performance for the three consecutive approach, which was W (70.9, 77.3), N1 (48.8, 64.3), N2(66.7, 92.2), N3(58.4, 94.4), R (69.1, 91.3).

	3 Co	onsecuti	ve 6 sec	-sub epo	och Met	hod				3 Noi	1-Conse	cutive	6 sec-su	b epoch	Method	
		Group	o1: Prin	nary inso	omnia						Grou	ıp1: Pr	imary ii	isomnia		
			Auto	matic det	ection							Au	tomatic o	letection		
		W	N1	N2	N3	REM	Sensitivity (%)				W	N1	N2	N3	REM	Sensitivity (%)
tion	W	1460	267	308	16	7	70.9		on al	W	1746	135	161	15	1	84.8
/isu	N1	200	450	116	150	43	48.8		isua ecti	N1	81	661	160	53	4	68.83
de	N2	558	310	2518	320	75	66.7		V det	N2	238	125	3316	99	3	87.7
	N3	191	354	123	982	31	58.4			N3	64	167	211	1234	5	73.4
	REM	82	145	10	20	576	69.1			REM	80	30	1	0	722	86.6
	Specificity (%)	77.3	64.3	92.2	94.4	91.3				Specificity	89.9	89.8	96.7	97.4	98.8	
	kappa				0.50					kappa				0.75		
	Agree (%)				64.2					Agree (%)				82.4		
		Gro	up2: Sc	hizophro	enia						Gre	oup2:	Schizop	hrenia		
			Auto	matic det	ection							Au	tomatic o	letection		
c c		W	N1	N2	N3	REM	Sensitivity (%)				W	N1	N2	N3	REM	Sensitivity (%)
ual	W	840	374	57	71	35	61.2		sual	W	1086	189	37	45	20	82.2
Vis letec	N1	134	534	47	43	22	65.2		/ist	N1	133	584	30	26	7	75.8
ŕ p	N2	1167	264	2477	255	57	59.4		de	N2	551	63	3522	81	3	85.2
	N3	136	24	68	1558	6	78.4			N3	31	2	48	1711	0	85.2
	REM	149	56	6	5	1414	91.8			REM	45	2	1	0	1582	97.1
	Specificity (%)	88.9	93.6	68.9	82.8	84.3				Specificity (%)	89.8	97.2	2 97.8	98.4	99.7	
	kappa				0.62					kappa				0.82		
	Agree (%)				71					Agree (%) 86.6						
		(	Froup3:	Healthy	/						G	roup3	Health	y		
			Auto	matic det	ection	-						Auto	matic de	tection		
u		W	N1	N2	N3	REM	Sensitivity (%)		_ u		W	N1	N2	N3	REM	Sensitivity (%)
sual	W	1219	703	57	194	66	52.8		sua ctic	W	1664	358	46	154	17	74
Vis ete	N1	151	543	72	71	43	59.2		Vi: lete	N1	183	625	36	28	8	68.7
р	N2	598	415	2638	406	78	64.2		q	N2	498	161	3316	154	6	80.8
	N3	35	9	63	1134	3	91.1			N3	5	1	41	1197	0	96.2
	REM	58	46	5	3	1130	91.2			REM	22	5	0	1	1214	97.7
	Specificity	84.1	93.2	67.83	86.9	87.9				(%)	89.4	93	97.3	95.7	99.4	
	kanna	07.1	15.4	07.05	0.6	07.9				kappa			~	0.74	~~··	
	Agree (%)				68.35					Agree (%)				80.85		
								-								

Table 15: The overall performance and comparison between visual and automatic detection (by both 3 consecutive and non-consecutive sub-epoch approaches) for the three groups.

#### 6.3 Individual subject performance

To obtain a view of the performance per subject, Cohen's Kappa coefficients were computed for the automatic sleep stage detection based on the 30-second epoch, three consecutive 6-second sub-epoch and three non-consecutive 6-second sub-epoch methods for each group of patients: Primary insomnia (Figure 16), schizophrenia (Figure 17) and healthy (Figure 18). For all three groups it can be seen from the Cohen's Kappa coefficients that automatic detection (by the three non-consecutive 6 second sub-epoch method) agrees substantially better with the visual detection (expert) than with the automatic detection by the three non-consecutive 6-second sub-epoch and 30-second approaches).



Figure 15: Hypnogram of visual sleep stage scoring versus automatic sleep stage scoring using 30-second epoch, three consecutive 6-second sub-epoch and three non-consecutive approaches, respectively.



Figure 16: Cohen's Kappa (C.K.) and accuracy (ACC) agreement of automatic detection approaches using 30-secend epoch, three consecutive 6-second and three non-consecutive 6-second methods with visual detection by expert for each primary insomnia patient (n=9 subjects, Sub 1-9).



Figure 17: Cohen's Kappa (C.K.) and Accuracy (ACC) agreement of automatic detection approaches using 30-second epoch, three consecutive 6-second and three non-consecutive 6-second methods with visual detection by expert for each schizophrenia patient (n = 10 subjects, Sub 1-10).



Figure 18: Cohen's Kappa (C.K.) and Accuracy (ACC) agreement of automatic detection approaches using 30-second epoch, three consecutive 6-second and three non-consecutive 6-second methods with visual detection by expert for each healthy subject (n = 10 subjects, Sub 1-10).

#### 6.4 Automatic detection of sleep stages using EOG

In this study, the EOG signal was utilised for the detection of the sleep stages of 30 patients, comprising 10 healthy (controls), 10 PLMS, and 10 SAHS. Several features were extracted from the EOG signal based on different frequency bands, as mentioned in the previous section (refer to section 4.3.2.2). The overall agreement, sensitivity and specificity of the sleep stage detection for healthy (control) subjects were 83.5%, 85%, and 88% respectively. The Cohen's Kappa was 0.79. Table 16 shows the confusion matrix with the sensitivity and specificity after applying the smoothing rule 1, for a single healthy subject. The results showed that the best detection was in W, and N3 (the sensitivity was 91%). The detection of sleep stage N1, after utilising the smoothing rule, was significantly improved. The overall agreement, sensitivity, and specificity for the detection of the sleep stages of the PLMS patients was 80%, 82%, and 86%, respectively. The Cohen's Kappa was 0.71, which was lower than the Cohen's Kappa for the healthy (controls) subjects. The reason for this is that the normal distribution of sleep stages with healthy subjects was much more consistent than with the PLMS patients. Table 17 shows the confusion matrix, sensitivity and specificity of the sleep stages for a single PLMS patient. It is clear that the total number of sleep stage N2s was higher than the other sleep stages; therefore, the detection of stages N1 and R was slightly lower than other stages. On the other hand, the overall agreement, sensitivity, and specificity for the detection of the sleep stages with the SAHS patients were 78%, 77%, and 80%, respectively, whilst the Cohen's Kappa was lower than in the other two groups (healthy and PLMS) by 0.67. Table 18 shows the confusion matrix, sensitivity and specificity of the sleep stages for a single SAHS patient. It is clear that the lower sensitivity was in the wakefulness stages, with an improvement in the detection of sleep stage N1.

Table 16: The confusion matrix of a single healthy (control) subject.

				Automa	tic detect	ion	
		W	N1	N2	N3	R	Sensitivity
							(%)
	W	575	18	120	5	16	91
	N1	22	238	108	5	29	78
Viewal	N2	15	10	2280	39	62	80.6
visuui detection	N3	3	1	168	1018	46	91
uciection	R	11	3	150	43	1065	87.4
	Specificity (%)	90	85	70	87.4	81.2	

Table 17: The confusion matrix of a single PLMS patient.

		Automatic detection									
		W	N1	N2	N3	R	Sensitivity				
							(%)				
	W	524	6	163	5	28	87				
	N1	50	150	163	9	36	70				
Vienal	N2	25	5	3091	13	44	80				
detection	N3	10	0	191	224	19	85				
	R	186	11	277	16	836	78				
	Specificity (%)	88	88	60	87.6	82.2					

Table 18: The confusion matrix of a single SAHS patient.

		Automatic detection										
		W	N1	N2	N3	R	Sensitivity					
							(%)					
	W	547	10	153	17	43	66					
Viewal	N1	29	134	181	11	59	72					
Visuai	N2	38	13	2893	68	138	79.5					
aelection	N3	3	2	104	731	24	84					
	R	11	12	255	32	968	85.7					
	Specificity	85	75	66	96	85						
	(%)											

Figures 19, 20 and 21 show the hypnograms of visual sleep stage scoring vs. automatic scoring for a healthy (control), a PLMS and an SAHS patient, respectively. It can be observed that some sleep stages were scored as sleep stage N2 or N3 which made the hypnogram included some incorrect classifications. Figures 22, 23 and 24 show the accuracy of the selected features for the detection of the sleep stages for each group.



Figure 19: The hypnogram of visual sleep stage scoring vs. automatic scoring for a single healthy (control) subject.



Figure 20: The hypnogram of visual sleep stage scoring vs. automatic scoring for a single PLMS patient.



Figure 21: The hypnogram of visual sleep stage scoring vs. automatic scoring for an SAHS patient.



Figure 22: The accuracy of the selected features for the detection of sleep stages for healthy subjects.



Figure 23: The accuracy of the selected features for detection sleep stages for PLMS patients.



Figure 24: The accuracy of the selected features for the detection of sleep stages for SAHS patients.

#### 6.5 Classification of insomnia, schizophrenia and healthy sleep

Figure 12 shows the automatic classification algorithm used to identify the patients with primary insomnia, schizophrenia and the healthy (control) patients. The sleep parameters of SL, the first REM period and sleep stage N1 were used to classify these three groups on the basis of the thresholds as described in the previous section (refer to section 5.1.2.2). Table 19 shows the post-hoc *t*-test analysis for the three groups of patients. There were significant differences (p = 0.01 and 0.00 respectively) between the primary insomnia patients and the healthy (control) subjects, particularly in sleep stages N1(%) and R(%). Moreover, there were significant differences between the schizophrenia patients and the healthy (control) subjects in the sleep parameters of WASO (p = 0.05), NW (p = 0.05) SWS duration (p = 0.04). The primary insomnia significantly differed from the schizophrenia patients in some sleep parameters, such as SL (p = 0.04), WASO (p = 0.00), NW (p = 0.00) and the REM duration (p = 0.00). Figure 25 shows the bar plots of the sleep parameters of the three patient groups. The

SL, NW, first REM period (FRP) and R sleep stages were all significant sleep stage parameters for the schizophrenia patients. Sleep stage N1(%) was a significant sleep parameter for the primary insomnia patients. The sensitivity and specificity of the identification the primary insomnia patients were 90% and 95%, respectively. For the schizophrenia patients, it was 90% and 81%, respectively. For the healthy (control) subjects it was 70% and 99 %, respectively (see Table 20). The level of accuracy and Cohen's Kappa was 83.4 % and 0.75, respectively.

Sleep parameters	Prim	Primary insomnia vs.			ary insomn	ia vs.	Schizophrenia vs. healthy		
		healthy		S	chizophren	ia			
	t	SD	р	t	SD	р	t	SD	Р
Sleep Latency	-0.1	26	0.85	2.2	18.6	*0.04	1.3	27.8	0.21
Sleep efficiency	0.7	16.5	0.47	1.6	6.6	0.13	1.4	16.1	0.18
WASO	-0.6	73.9	0.55	-4.2	21.2	*0.00	-2.1	64.4	*0.05
Number of	-0.6	148.4	0.50	-3.4	48.8	*0.00	-2.1	128.8	*0.05
awakenings									
REM latency	0.4	56.2	0.67	-0.1	81.4	0.87	0.1	77.7	0.88
First REM period	-1.	5.8	0.08	1.6	8.9	0.13	0.3	10.3	0.76
Total sleep time	1.1	80.2	0.28	1.3	38	0.21	1.9	74.2	0.08
SWS duration	1.8	59.5	0.10	-0.1	38	0.88	2.3	43.3	*0.04
<b>REM</b> duration	-1.9	28.3	0.07	4.7	24	*0.00	1.7	34.5	0.10
N1 (%)	-7.7	20.2	*0.01	-0.8	10	0.43	-0.6	11.8	0.50
N2 (%)	-0.5	13.2	0.57	-0.9	8.6	0.34	-2	7.7	0.06
N3 (%)	1.7	13.9	0.11	-0.5	16	0.61	8.9	8.1	0.28
R (%)	-3.1	5.6	*0.00	4.3	6.3	*0.00	1.3	6.8	0.20
N1/N2	-0.8	24.5	0.41	-0.6	19.2	0.53	-1.1	28.5	0.26
N1/N3	-0.12	48.1	0.9	0.2	38.1	0.78	0.1	34.2	0.88
N1 (%) N2 (%) N3 (%) R (%) N1/N2 N1/N3	-7.7 -0.5 1.7 -3.1 -0.8 -0.12	20.2 13.2 13.9 5.6 24.5 48.1	*0.01 0.57 0.11 *0.00 0.41 0.9	-0.8 -0.9 -0.5 4.3 -0.6 0.2	10 8.6 16 6.3 19.2 38.1	0.43 0.34 0.61 *0.00 0.53 0.78	-0.6 -2 8.9 1.3 -1.1 0.1	11.8 7.7 8.1 6.8 28.5 34.2	0.50 0.06 0.28 0.20 0.26 0.88

Table 19: The post-hoc *t*-tests of the differences in the sleep parameters between the three groups.

\* *p* < 0.05.

Table 20: Confusion matrix of detection of the three participant groups based on the decision three analysis.

	Automatic classification										
		Healthy	Sensitivity (%)								
		insomnia									
	Primary	9	1	0	90%						
	insomnia										
True	Schizophrenia	1	9	0	90%						
classification	Healthy	0	3	7	70%						
	Specificity (%)	95%	81%	99%							
	Cohen's Kappa	0.75									



Figure 25: Bar plots of the different sleep parameters for the primary insomnia patients, schizophrenia patients and the healthy subject. SWD = slow-wave sleep duration; FPR = the first period REM; RED = the REM duration.

#### 6.6 Classification of PLMS, SAHS and healthy sleep

Figure 13 shows the automatic classification algorithm used to classify the PLMS and SAHS patients, and the healthy (control) subjects. The significant sleep parameters of N1(%), N2(%), and SWSD were used to identify the three groups on the basis of the thresholds, as described in the previous section (see section 5.2.2.2).

Table 21 shows the post-hoc *t*-test analysis for the three groups of patients. There were significant differences between the PLMS patients and the healthy control participants, particularly in sleep stages N2(%) (p = 0.01), N3(%) (p = 0.01) and SWS(%) (p = 0.03) duration. Furthermore, there were significant differences between the SAHS patients and the healthy (control) subjects in the sleep parameters of SWS duration (p = 0.01),

REM duration (p = 0.01), and sleep stages N2 (%) (p = 0.01), N3 (%) (p = 0.01), and

R (%) (p = 0.01).

Table 21: The post-hoc *t*-tests of the differences in the sleep parameters between the three patient groups.

	PLMS vs Healthy		PLMS vs SAHS			SAHS vs Healthy			
Sleep parameters	t	SD	р	t	SD	р	t	SD	р
Sleep latency	0.37	43.2	0.71	0.2	40.2	0.81	0.28	21.5	0.77
Sleep efficiency	-0.62	22.4	0.54	-0.57	18.3	0.57	-0.24	14.5	0.81
WASO	0.61	94.9	0.55	0.50	81.04	0.62	0.29	59.6	0.77
Number of wakings	0.58	1.13	0.57	0.48	193	0.64	0.28	143	0.78
REM latency	1.9	48.36	0.07	-0.43	101.7	0.67	1.4	95.2	0.17
First REM period	-2.07	0.08	0.06	-1.0	0.60	0.34	0.69	0.60	0.5
Total sleep time	-0.41	118.9	0.68	-0.51	90.6	0.61	-0.03	77.3	0.97
SWS duration	-2.56	70.5	*0.03	50.2	0.51	0.62	-6.01	34.38	*0.01
<b>REM</b> duration	-1.8	41.7	0.10	1.0	32.9	0.33	-3.25	33.4	*0.01
W (%)	0.30	23.2	0.76	0.30	18.8	0.76	0.08	15.38	0.93
N1 (%)	2.07	6.1	0.06	1.2	5.7	0.25	1.34	4.23	0.21
N2 (%)	2.95	16.6	*0.01	-2.93	10.2	*0.01	5.25	15.06	*0.01
N3 (%)	-3.04	15.2	*0.01	0.80	11.7	0.44	-7.61	7.30	*0.01
R (%)	-1.38	10.3	0.2	1.75	6.1	0.11	-2.78	9.03	*0.02
* .0.05									

\* p < 0.05.

Table 22: Confusion matrix of detection of the PLM and SAHS patients and healthy subjects based on the decision three analysis.

Automatic classification								
		PLMS	SAHS	Healthy				
	PLMS	9	1	0				
Truc	SAHS	0	9	1				
lassification	Healthy	1	0	9				
classification	Sensitivity (%)	90	90	90				
	Specificity (%)	95	95	95				
	Cohen's Kappa		0.85					

The PLMS differed from the SAHS patients within the sleep parameters of sleep stage N2 (%) (p = 0.01). Figure 26 shows the bar plots of the three groups. The SL, WASO, and NW were all significant sleep stage parameters for the PLMS patients. Sleep stage N2 was a significant sleep parameter for the SAHS patients. The sensitivity and specificity of identification in the PLMS patients was 90% and 95%, respectively, the SAHS patients was 90% and 95%, respectively, and the healthy (controls) was 90%

and 95%, respectively (see Table 22). The level of accuracy and Cohen's Kappa were 90% and 0.85, respectively.



Figure 26: Bar plots of the different sleep parameters for the PLMS and SAHS patients and healthy subjects.

### 7 : Discussion

## 7.1 Automatic detection of wake and sleep stage N1 using the EEG sub-epoch approach

The results of this study confirmed the possibility of using the 30-second epoch and the three consecutive or non-consecutive 6-second epoch methods for the scoring of W and stage S1 sleep. Whilst the accuracy varied between subjects, the non-parametric method proved to be more effective with stage S1 sleep detection, whereas the parametric method was more effective for the W stage detection. The non-consecutive sub-epoch method was more effective and the three consecutive method was least effective in the non-parametric stage S1 detection. The 30-second epoch method was most effective for the parametric W stage detection.

Parametric and non-parametric methods with approach of consecutive and nonconsecutive sub-epochs, using EEG signals to evaluate PSG (napping) database can contribute to improving the discrimination between W and sleep S1. These methods can also be used in evaluating the excessive daytime sleepiness (EDS) conditions.

However, there were limitations in this pilot study. There is a need to design an adaptive detector that is intelligent enough to know when to apply the non-parametric and parametric 30-second and the consultative/non-consecutive sub-epoch methods. Accuracy was calculated independently for W and S1, rather than combined. The PSG data, recorded without biofeedback conditions, would process different results, which subjects would exhibit non-induced W and S1 EEG activity. Other improvements to these results may be found in including relative EEG delta power. Also, the frequency bands that were used in this study were based on the study of Diego [178]. Therefore, further improvement may occur in the detection of S1 if the frequency band of theta to

be between 4 to 8 Hz. Moreover, using the smoothing rule that was described in the previous section (see section 4.2.2) may also improve the accuracy of detection W and S1, which was not applied in this study. Also, the pre-processed step of filtering may also need to be excluded to improve the EEG bands power computations.

#### 7.2 Automatic detection of sleep stages using EEG sub-epoch approach

In this study, the main aim was to detect the sleep stages based on three consecutive and non–consecutive sub-epochs by using a single channel EEG (Cz location). The algorithm performance was reliable and satisfactory across all subjects. The overall agreement between the manual and automatic scoring was 67.85% for the three consecutive 6- second sub-epoch method, and 83.3% for the three non-consecutive 6second sub-epoch method. The reason that an overall accuracy was low (see Table 15) for the three consecutive 6-second sub-epoch method, is that some 30-second epoch were not included in three consecutive sub-epoch of any sleep stages.

It is challenging to discriminate the sleep stage R from N1 or W stage with an automatic sleep detection approach [195,196] because of the criteria that scoring R sleep is based on, the EOG signal (present REM), and EMG (muscle atonic) according to the AASM standard. The spectral analysis of the EEG signal showed the frequency mixture throughout the R stage with delta and sigma band power [197, 198]. The combination between these frequencies has been reported as a distinctive EEG marker of R sleep [197]. In order to distinguish between R sleep and other sleep stages, the three consecutive and non-consecutive sub-epoch was utilised with the band power of saw-tooth (2-6 Hz) or a combination between the delta2 (2–4 Hz), sigma (12–15 Hz), beta1 (13–20 Hz) and beta2 (20–30 Hz).
Table 23 compares the three consecutive and non-consecutive 6-second sub-epoch approach with four commercial sleep stage detection algorithms of FP-STAGER [180], ASEEGA [197], ZEO [198] and ARES [199]. These algorithms all use a single EEG signal and were compared by epoch-by-epoch manual scoring. The candidature's own 3 consecutive and non-consecutive 6-second sub-epoch approach with a single EEG (Cz channel location) was also compared with the expert epoch-by-epoch scoring, as shown in Table 23.

References	Number	Signal EEG	Sensitivity	Specificity	Agreement	Cohen's
	of	C	(%)	(%)	(%)	Kappa
	subjects					
[180]	29	EEG (Fp1-	-	-	80.0	0.75
		Fp2)				
[197]	15	EEG (Cz-Pz)	82.5	-	82.9	0.72
[198]	26	EEG (Fp1-	-	-	84.0	0.57
		Fp2)				
[199]	20	EEG (Fp1-	-	-	83.2	0.74
		Fp2)				
Candidature	29	EEG (Cz)	82.9	95.2	83.3	0.77
result						

Table 23: A comparison of candidature own approach with existing commercial algorithms based on the single EEG channel.

*Note.* - = not measured.

Candidature results indicated that the detection of W stage with some subjects (in Group 3) had low accuracy (74 %), due to a very low alpha activity over Cz location, which did not satisfy the criteria of scoring the W stage. Other algorithms performed similarly [199]. Additionally, most algorithms [182, 200] showed a very low detection accuracy for sleep stage N1. However, the three consecutive and non-consecutive 6-second sub-epoch (EEG 'Cz') approach yielded improved detection of sleep stage N1.

The candidature attempted to compare the performance of the automatic detection of sleep stages based on 30-second epoch (prior to dividing the 30-second epoch into sub-epoch approach). There was a significant improvement when utilising the three consecutive and non-consecutive sub-epoch in comparison to the 30-second epoch.

This study had some limitations. Firstly, this approach was validated on a small population of young healthy and non-healthy subjects. Therefore, additional studies on elderly healthy and non-healthy subjects is necessary to fully evaluate this algorithm. Secondly, the features used in this algorithm were selected based on the optimal features for each sleep stage that were described in previous studies [153,188,189]. An algorithm can be optimised by selecting the best features for detecting each sleep stage. Finally, the sleep stage rules where optimized for this PSG data and thus the algorithm presented here may be suboptimal if utilised with different sleep stages rules or data.

## 7.3 Automatic detection of sleep stages using EOG and classification of PLMS, SAHS and healthy sleep

This pilot study used EOG signals for automatic sleep stage detection, and then used the data (sleep stages that detected automated to measure sleep parameters) to classify PLMS and SAHS patients and healthy (control) subjects. The overall inter-rater agreement between the visual and automatic sleep stage scoring for the three groups was 80.5%, with a Cohen's Kappa of 0.73. On the other hand, the agreement level of the automatic classification of sleep disorders was 90%, and Cohen's Kappa was 0.85.

Different features were employed extracted from the EOG signals and then fed into the KNN classifier for detection of the wake and the sleep stages. Some studies have used the decision rule-based approach with various thresholds to predict the sleep stages [137]. However the results indicated a slight improvement of the specific EEG alpha thresholds for offline applications, with an accuracy of 73%. Therefore, the present study used the KNN classifier due to its simplicity and strength in detecting the sleep stages. Several studies have employed signals in addition to EOG signals for automatic

sleep stage detection, such as EEG and EMG signals [150,151,178]. These require more electrodes and more complicated algorithms to increase the accuracy level. On the other hand, some studies used only one EEG signal for automatic sleep detection [176,201]. Since the predominance of sleep stage N2 in PLMS and SAHS patients was more than in the healthy (control) subjects, this was a distinct difference between these three groups. This led to the overall accuracy of the sleep stages of the PLMS and SAHS patients to be very low (80% and 78% respectively), which meant that the KNN classifier detected the other sleep stages as sleep stage N2. In Table 17, for example, it was obvious that the total number of occurrences of sleep stage N2 was higher than in the other sleep stages, which caused an increased overall detection of the other sleep stages or the wakefulness stage.

Similar studies have utilised the EOG signal for the detection of the sleep stages, or of one particular sleep stage such as SWS (N3) [136,201]. An automatic method was previously developed for detection of SWS based on two EOG channels [136]. This study employed the amplitude criterion for detecting SWS, and beta power [18-30] was utilised to reduce the artefact. The result showed inter-rater agreement between the visual and the developed automatic method was 93%, with a Cohen's Kappa value of 0.70. The sensitivity and specificity was 75% and 96%, respectively. Another study employed two-channel EOG with the reference to the left mastoid (M1) for the automatic sleep stage detection [136]. The synchronous EEG activity during S2 and S3 was detected by calculating peak-to-peak and cross-correlation amplitude differences in the 0.5-6 Hz range (between the two EOG channels). The result indicated that the epoch-by-epoch agreement between the visual scoring and the automatic method was 72%, with a Cohen's Kappa value of 0.63.

The second aim of this study was to utilise the automated detection of sleep stages for the purpose of classifying PLMS and SAHS patients, and healthy (control) subjects. The results provided evidence to support the use of the PSG sleep stage parameters, such as sleep stage N1(%), N2(%), and SWSD in order to automatically classify PLMS and SAHS patients, and healthy (control) subjects. The stage N1 (%) was the most significant parameter distinguishing the SAHS patients from the healthy (control) subjects. The study found that sleep stage N1 (%) was 9% for seven SAHS patients. Conversely, eight healthy subjects had a sleep stage N1 (%) for less than 5% of the total sleep duration. However, some patients with PLMS showed higher sleep stage N1 (%), which led the mean average of sleep stage N1 (%) to be higher than in the other two groups. Figure 26 presents evidence that the mean average of sleep stage N1 (%) for the healthy (control) group was lower compared with the other groups (PLMS and SAHS).

The sleep stage N2 (%) was used to distinguish between the SAHS and PLMS patients and healthy (control) subjects. This study found that most of the SAHS patients had a higher sleep stage N2 (%) (above 80%) than the PLMS patients. Figure 13 shows the threshold that was used to successfully distinguish between the SAHS patients and the other two groups (PLMS and healthy). The SWSD and sleep stage N3 (%) was used to discriminate the PLMS patients from the healthy (control) subjects. It was found that most of the PLMS patients had shorter SWSDs compared to the healthy subjects. The longest SWSD of the healthy (control) group was above 70 minutes. The reason for using sleep stage N3 (%) was because some PLMS patients had similar SWSDs to the healthy (control) group. The overall accuracy was 90%, and the Cohen's Kappa was 0.85.

#### 7.4 Classification of insomnia, schizophrenia and healthy sleep

This pilot study provided evidence to support the use of the PSG sleep stage parameters, such as SL, FRP, WASO and sleep stage N1(%) for an automatic classification of primary insomnia and schizophrenia. The first REM period was the most significant parameter to distinguish between the primary insomnia and schizophrenia patients. The study found that this period was less than two minutes for eight primary insomnia patients. Conversely, eight schizophrenia patients showed a first REM period of 2 to 10 minutes. Figure 25 presents evidence that the mean average duration of the first REM period for the primary insomnia group was lower compared with the other groups (schizophrenia and healthy). Therefore, this parameter was used to discriminate the primary insomnia from the schizophrenia patients. To avoid misclassifications between the primary insomnia and schizophrenia patients, the sleep stage N1 (%) parameter was applied, as patients with primary insomnia showed an increase in sleep stage N1 (%) [54]. Figure 12 shows that the threshold used for addressing this problem was above 12.5% N1 (%). Therefore, the sensitivity in identifying primary insomnia from schizophrenia was 90%, which meant that nine patients from each group were accurately identified. The SL parameter was used to distinguish between the schizophrenia patients and healthy subjects since most studies have shown that patients with schizophrenia have a long SL [202]. Here, seven out of ten healthy (controls) subjects were perfectly classified and three patients were identified as schizophrenia patients, as shown in Table 20. The reason for this might be that, in relation to the schizophrenia patients, these three participants had the first degree of the schizophrenia condition. In order to investigate the sleep quality parameters, a post-hoc t-test analysis was used. There was a significant difference between the primary insomnia patients and healthy subjects in sleep stages N1 (%) and R(%). The

schizophrenia patients varied from the healthy subjects in some sleep parameters, such as WASO, NW and the SWSD. However, this study found that there was no significant difference between the schizophrenia patients and healthy subjects in the R(%) sleep over the total sleep stages, which could be due to the effects of the monotherapy (second generation antipsychotic). Also, while some studies have confirmed that patients with schizophrenia seemed to exhibit regular R(%) sleep stage [203, 204], the schizophrenia patients in this study showed a significant discriminate with the primary insomnia patients in the sleep parameters of SL, WASO, NW and the RED.

# 8 : Conclusion and future work

#### 8.1 Conclusion

To address the abnormal phenomena that occurs during the process of sleep, a proper diagnosis by a sleep physician is required. Firstly, the aim of this thesis was to focus primarily on the evaluation of the macrostructure of sleep and the detection of the sleep stages. Secondly, the information from the sleep stage feature extracts were applied in the process of classifying various sleep disorders and healthy (control) . In this thesis, two types of PSG data were used involving the short-time PSG recording (napping) and long-time PSG recording (overnight). For these PSG recordings, multiples electrodes were utilised which often cause to sleep disturbance. Therefore, this thesis used a single electrode for the recording of the EOG or EEG signal for the detection of the sleep stages. The first type of data included only the W and N1 sleep stages. These two stages are very important for the assessment/evaluation of the EDS condition. Two techniques were employed the parametric and non-parametric methods in order to evaluate the short-time data PSG recording and analysis of the napping data. The results of this study provided evidence of the effectiveness of these methods to successfully detect the W and N1 sleep stages.

Most studies have had difficulty implementing automated detection methods to distinguish between sleep stages, in particular the sleep stages, N1 and R [14,153]. This thesis aimed to effectively overcome this problem, by applying the consecutive and non-consecutive EEG sub-epoch approach. The results confirmed the benefit of automatic sleep stage scoring based on 6-second sub-epochs. In particular, the non-consecutive sub-epoch method was a favourable method for the detection of all sleep stages and for all groups. The application of this method could be used to improve the detection of sleep stages in ambulatory sleep monitoring conditions.

In this thesis, visual sleep stage scoring information (measured sleep quality parameters) was used for the automatic classification of primary insomnia, schizophrenia and healthy (control) subjects. The results have provided evidence to support the hypothesis that the PSG sleep stage parameters, (i.e. SL, FRP, WASO and sleep stage N1) can successfully be applied to automatically classify primary insomnia and schizophrenia. The results indicated that the first REM period was the most significant sleep stage parameter used to distinguish between primary insomnia and schizophrenia patients. The results have shown that the sensitivity of identifying primary insomnia, schizophrenia and healthy (control) subjects was high.

Furthermore, this thesis aimed also, to develop an automatic method for the detection of the sleep stages based on EOG signals, and then utilised these sleep stage parameters for the classification of the PLMS and SAHS patients, and healthy (control) subjects. There is a significant advantage that supports the use of automatic sleep stage detection based on only EOG signals for future ambulatory sleep monitoring. The sensitivity of identifying PLMS and SAHS patients and healthy (control) was very high. This suggests that using an automatic classification system in screening processes is more effective and efficient compared to some standards, such as the Pittsburgh Sleep Quality Index (PSQI) [205].

In terms of publications, it can be summarised that each of the publications has addressed each of the aforementioned research aims (section 1.3) and have made a contribution to the body of knowledge.

Here, citations refer to the previous section (see List of Publications). The author's following papers (2) (3) (4) present a new approach for automatic sleep stage detection based on the sub-epoch of the EEG signal. Instead of utilising multiple electro-

117

physiological signals, such as EEG, EOG and EMG for the detection of the sleep stages, a single EEG signal alone can provide a significant advantage to reduce the sleep disturbance caused by multiple electrode connections and to simplify the training and eliminate the reliance on sleep experts. In addition, the classification of the sleep stages based on the consecutive or non-consecutive sub-epoch approach led to an increase in the degree of successfully distinguishing between the sleep stages. Through these publication research aims 1(a), 1(b), 1(c) and 1(d) (refer to section 1.3) are achieved and the results are validated. Thus, these provide a contribution to the body of knowledge.

The author's following papers (1) (5) (6) (see List of Publications) address the challenge in the detection of the sleep stages based on the EOG signals only. This work can be applied in the future development of ambulatory sleep monitoring by using two or four electrodes for an automatic sleep stages detection. Several features were extracted from the EOG signals to improve the detection of the wakefulness and sleep stages. The results showed a strong possibility of detecting the sleep stages using only EOG signals. Through these publications, the research aims 1(e) and 1(f) (refer to section 1.3) are achieved and the results are validated.

All these publications individually show the positive optimal outcome meeting the second research aim (develop an automatic classification of sleep disorder (refer to section 1.3))

The author's following paper (1) (7) (see List of Publications) presents a new approach for the automatic classification of sleep disorders. The existing methods rely on the Pittsburgh Sleep Quality Index (PSQI), as a standardised subjective measure to evaluate sleep quality. PSQI is based on several questions relating to the evaluation of sleep quality for a duration of one month. This is a tedious and time-consuming task for sleep physicians. The best solution for this problem is to develop an automatic system based on PSG sleep stage parameters in order to classify different sleep disorders. Consequently, the results provide significant evidence to support the use of the PSG sleep stage parameters in the classification of different sleep disorders, thus assisting the sleep physician during the screening processes. Hence, through this publication, research aims 2(g) and 2(h) (refer to section 1.3) and results have been validated.

#### 8.2 Future Work

In order to know when to use the parametric and/or non-parametric method, there is a need to develop an intelligent adaptive filter. Also, the development of new sleep stage rules for different patient age groups and sleep disorders is recommended for future work. Moreover, to extract more features from the EEG and EOG signals is recommended for future work to improve the detection of the sleep stages.

It is recommended to develop rules for detection of sleep stage based on the R&K manual scoring. Furthermore, it is recommended to utilise the algorithm of consecutive and non-consecutive sub-epoch EEG for detection of sleep microstructure events such as various arousals. Moreover, it is recommended to evaluate for the effectiveness of the consecutive and non-consecutive sub-epoch approaches with other PSG signals such as EOG and EMG in order to detect the sleep stages. In order to compare the accuracy and speed of detection sleep stage with respect to the consecutive and non-consecutive sub-epoch EEG algorithm, other classification systems such as artificial neural network (AAN) and support vector machines (SVM) can be tested.

Also, there is a need to extend the range of parameters and rules for the classification of sleep disorders based on sleep quality. This thesis limitation was based on the validation of an algorithm on a small population of young healthy and non-healthy subjects; there is a need for future studies on elderly healthy and patients, with a large sample size.

### REFERENCES

- T. Lee-Chiong, *Sleep medicine: Essentials and review*: Oxford University Press, 2008.
- [2] E. Mignot, "Why we sleep: the temporal organization of recovery," *PLoS Biol*, vol. 6, p. e106, 2008.
- [3] R. Stickgold and M. P. Walker, "Memory consolidation and reconsolidation: what is the role of sleep?," *Trends in Neurosciences*, vol. 28, pp. 408-415, 2005.
- [4] G. Tononi and C. Cirelli, "Sleep function and synaptic homeostasis," *Sleep Medicine Reviews*, vol. 10, pp. 49-62, 2006.
- [5] K. L. Knutson, K. Spiegel, P. Penev, and E. Van Cauter, "The metabolic consequences of sleep deprivation," *Sleep Medicine Reviews*, vol. 11, pp. 163-178, 2007.
- [6] A. A. Prather, D. Janicki-Deverts, M. Hall, and S. Cohen, "Behaviorally assessed sleep and susceptibility to the common cold," *Sleep*, vol. 38, p. 1353, 2015.
- [7] J. Lim and D. F. Dinges, "Sleep deprivation and vigilant attention," *Annals of the New York Academy of Sciences*, vol. 1129, pp. 305-322, 2008.
- [8] T. J. Balkin, T. Rupp, D. Picchioni, and N. J. Wesensten, "Sleep loss and sleepiness: current issues," *CHEST Journal*, vol. 134, pp. 653-660, 2008.
- [9] P. Philip and T. Åkerstedt, "Transport and industrial safety, how are they affected by sleepiness and sleep restriction?," *Sleep Medicine Reviews*, vol. 10, pp. 347-356, 2006.
- [10] Anund, G. Kecklund, B. Peters, and T. Akerstedt, "Driver sleepiness and individual differences in preferences for countermeasures," *Journal of Sleep Research*, vol. 17, pp. 16-22, 2008.

- [11] M. Sallinen and G. Kecklund, "Shift work, sleep, and sleepiness—differences between shift schedules and systems," *Scandinavian journal of work, environment* & *health*, pp. 121-133, 2010.
- [12] Rechtschaffen and A. Kales, "A manual of standardized terminology, techniques and scoring system for sleep stages of human subjects," 1968.
- [13] o. S. Medicine and C. Iber, The AASM manual for the scoring of sleep and associated events: rules, terminology and technical specifications: American Academy of Sleep Medicine, 2007.
- [14] Y. Liu, L. Yan, B. Zeng, and W. Wang, "Automatic Sleep Stage Scoring Using Hilbert-Huang Transform with BP Neural Network," in *Bioinformatics and Biomedical Engineering (iCBBE), 2010 4th International Conference on*, 2010, pp. 1-4.
- [15] S. Uchida, M. Matsuura, S. Ogata, T. Yamamoto, and N. Aikawa, "Computerization of Fujimori's method of waveform recognition A review and methodological considerations for its application to all-night sleep EEG," *Journal of Neuroscience Methods*, vol. 64, pp. 1-12, 1996.
- [16] J. R. Smith, M. Negin, and A. H. Nevis, "Automatic analysis of sleep electroencephalograms by hybrid computation," *Systems Science and Cybernetics, IEEE Transactions on*, vol. 5, pp. 278-284, 1969.
- [17] M. Koivuluoma, A. Värri, and A. Flexer, "Modelling sleep with gaussian mixture model based on eye movements and delta-activity," *Signal Processing X Theories and Applications, Proceedings of EUSIPCO*, pp. 35-38, 2000.
- [18] A. Flexer, P. Sykacek, I. Rezek, and G. Dorffner, "Using hidden Markov models to build an automatic, continuous and probabilistic sleep stager," in *Neural Networks*,

2000. IJCNN 2000, Proceedings of the IEEE-INNS-ENNS International Joint Conference on, 2000, pp. 627-631.

- [19] D. A. Economics, "Re-awakening Australia: The economic cost of sleep disorders in Australia, 2010," *Sleep Health Foundation, Canberra*, 2011.
- [20] T. C. Andringa and J. J. L. Lanser, "How pleasant sounds promote and annoying sounds impede health: A cognitive approach," *International journal of Environmental Research and Public Health*, vol. 10, pp. 1439-1461, 2013.
- [21] V. M. Kumar, "Sleep is neither a passive nor an active phenomenon," *Sleep and Biological Rhythms*, vol. 8, pp. 163-169, 2010.
- [22] R. L. Sack, D. Auckley, R. R. Auger, M. A. Carskadon, K. P. Wright Jr, M. V. Vitiello, *et al.*, "Circadian Rhythm Sleep Disorders: Part I, Basic Principles, Shift Work and Jet Lag Disorders An American Academy of Sleep Medicine Review: An American Academy of Sleep Medicine Review," *Sleep*, vol. 30, p. 1460, 2007.
- [23] M. M. Weber and W. Burgmair, "The assistant's bedroom served as a laboratory: Documentation in 1888 of within sleep periodicity by the psychiatrist Eduard Robert Michelson," *Sleep Medicine*, vol. 10, pp. 378-384, 2009.
- [24] H. Blake and R. W. Gerard, "Brain potentials during sleep," American Journal of Physiology, 1937.
- [25]G. Patrick and J. A. Gilbert, "Studies from the psychological laboratory of the University of Iowa: On the effects of loss of sleep," *Psychological Review*, vol. 3, p. 469, 1896.
- [26] L. Loomis, E. N. Harvey, and G. Hobart, "Potential rhythms of the cerebral cortex during sleep," *Science*, 1935.
- [27] A. A. Borb and P. Achermann, "Sleep homeostasis and models of sleep regulation," *Journal of Biological Rhythms*, vol. 14, pp. 559-570, 1999.

- [28] M. V. Sanchez-Vives and D. A. McCormick, "Cellular and network mechanisms of rhythmic recurrent activity in neocortex," *Nature Neuroscience*, vol. 3, pp. 1027-1034, 2000.
- [29] R. J. Berger and I. Oswald, "Effects of sleep deprivation on behaviour, subsequent sleep, and dreaming," *The British Journal of Psychiatry*, vol. 108, pp. 457-465, 1962.
- [30] A. A. Borbély, F. Baumann, D. Brandeis, I. Strauch and D. Lehmann, "Sleep deprivation: effect on sleep stages and EEG power density in man," *Electroencephalography and Clinical Neurophysiology*, vol. 51, pp. 483-493, 1981.
- [31]S. M. Jay, N. Lamond and S. A. Ferguson. "The characteristics of recovery sleep when recovery opportunity is restricted," *Sleep*, vol. 30, pp. 353-360, 2007.
- [32] N. Pedersen, P. Fuller, J. Lu, and C. Saper, "In the flicker of an eye," *The Journal of Physiology*, vol. 586, pp. 3305-3306, 2008.
- [33] J. Dewar. "The physiological action of light", Nature, vol. 15, pp. 433-5,1877.
- [34] E. Jacobson, "Electrical measurements of neuromuscular states during mental activities. III. Visual imagination and recollection," *American Journal of Physiology*, 1930.
- [35]O. Mowrer, T. C. Ruch, and N. Miller, "The corneo-retinal potential difference as the basis of the galvanometric method of recording eye movements" *American Journal of Physiology--Legacy Content*, vol. 114, pp. 423-428, 1935.
- [36]E. Aserinsky and N. Kleitman, "Regularly occurring periods of eye motility, and concomitant phenomena, during sleep" *Science*, vol. 118, pp. 273-274, 1953.
- [37] I. Feinberg, T. Floyd, and J. March, "Effects of sleep loss on delta (0.3–3 Hz) EEG and eye movement density: new observations and hypotheses" *Electroencephalography and Clinical Neurophysiology*, vol. 67, pp. 217-221, 1987.

- [38] F. Lucidi, A. Devoto, C. Violani, L. De Gennaro, P. Mastracci, and M. Bertini, "Rapid eye movements density as a measure of sleep need: REM density decreases linearly with the reduction of prior sleep duration," *Electroencephalography and clinical neurophysiology*, vol. 99, pp. 556-561, 1996.
- [39] L. De Gennaro, M. Ferrara and M. Bertini, "The relationship between frequency of rapid eye movements in REM sleep and SWS rebound" *Journal of Sleep Research*, vol. 9, pp. 155-159, 2000.
- [40] G. Barbato, C. Barker, C. Bender, H. A. Giesen, and T. A. Wehr, "Extended sleep in humans in 14 hour nights (LD 10: 14): relationship between REM density and spontaneous awakening," *Electroencephalography and Clinical Neurophysiology*, vol. 90, pp. 291-297, 1994.
- [41]E. Aserinsky, "Rapid eye movement density and pattern in the sleep of normal young adults," *Psychophysiology*, vol. 8, pp. 361-375, 1971.
- [42] P. Ktonas, A. Nygren and J. Frost, "Two-minute rapid eye movement (REM) density fluctuations in human REM sleep," *Neuroscience Letters*, vol. 353, pp. 161-164, 2003.
- [43] M. J. Thorpy, "Classification of sleep disorders" Journal of Clinical Neurophysiology, vol. 7, pp. 67-82, 1990.
- [44] D. Hume, "Emotions and moods" Organizational Behavior, pp. 258-297, 2012.
- [45] C. Kushida, *Encyclopedia of Sleep*: Academic Press, 2012.
- [46] J. D. Edinger, M. H. Bonnet, R. R. Bootzin, K. Doghramji, C. M. Dorsey, C. A. Espie, *et al.*, "Derivation of research diagnostic criteria for insomnia: report of an American Academy of Sleep Medicine Work Group," *Sleep*, vol. 27, pp. 1567-1596, 2004.

- [47] National Institutes of Health, "National Institutes of Health State of the Science Conference statement on Manifestations and Management of Chronic Insomnia in Adults, June 13-15, 2005," *Sleep*, vol. 28, p. 1049, 2005
- [48] American Psychiatric Association, "Diagnostic and statistical manual of mental disorders," 1980. [Online].

Available:http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.471.4319& rep=rep1&type=pdf

- [49] C. M. Morin, "Definition of acute insomnia: diagnostic and treatment implications," *Sleep Medicine Reviews*, vol. 16, pp. 3-4, 2012.
- [50] G. Why, "International classification of diseases," 2015.
- [51]A. S. D. Association and D. Committee, "International classification of sleep disorders: diagnostic and coding manual" *American Academy of Sleep Medicine*, 2005.
- [52] R. L. Sack, D. Auckley, R. R. Auger, M. A. Carskadon, K. P. Wright Jr, M. V. Vitiello, *et al.*, "Circadian Rhythm Sleep Disorders: Part I, Basic Principles, Shift Work and Jet Lag Disorders. An American Academy of Sleep Medicine Review." *Sleep*, vol. 30, p. 1460, 2007.
- [53]K. Šonka, P. Šóš, and M. Susta, "Past and present in drug treatment of sleep disorders," *Neuroendocrinology Letters*, vol. 35, 2014.
- [54] American Psychiatric Association, "Diagnostic and statistical manual of mental disorders (DSM)," Washington, DC: American Psychiatric Association, pp. 143-7, 1994.
- [55] H. Moore IV, E. Leary, S.-Y. Lee, O. Carrillo, R. Stubbs, P. Peppard, *et al.*, "Design and Validation of a Periodic Leg Movement Detector," *PloS one*, vol. 9, p. e114565, 2014.

- [56] N. Vadivelu, A. Vadivelu, and A. D. Kaye, Orofacial Pain: A Clinician's Guide: Springer Science & Business Media, 2014.
- [57] M. Michaud, A. Chabli, G. Lavigne, and J. Montplaisir, "Arm restlessness in patients with restless legs syndrome," *Movement Disorders*, vol. 15, pp. 289-293, 2000.
- [58] M. Bonnet, D. Carley, M. Carskadon, P. Easton, C. Guilleminault, R. Harper, *et al.*,"Recording and scoring leg movements," *Sleep*, vol. 16, pp. 749-759, 1993.
- [59] G. Mbata and J. Chukwuka, "Obstructive sleep apnea hypopnea syndrome," *Annals of Medical and Health Sciences Research*, vol. 2, pp. 74-77, 2013.
- [60] A. I. Pack, G. Maislin, B. Staley, F. M. Pack, W. C. Rogers, C. F. George, *et al.*, "Impaired performance in commercial drivers: role of sleep apnea and short sleep duration," *American journal of respiratory and critical care medicine*, vol. 174, pp. 446-454, 2006.
- [61]L. G. Sulit, A. Storfer-Isser, C. L. Rosen, H. L. Kirchner, and S. Redline, "Associations of obesity, sleep-disordered breathing, and wheezing in children," *American journal of Respiratory and Critical Care Medicine*, vol. 171, pp. 659-664, 2005.
- [62] J. M. Marin, S. J. Carrizo, E. Vicente, and A. G. Agusti, "Long-term cardiovascular outcomes in men with obstructive sleep apnoea-hypopnoea with or without treatment with continuous positive airway pressure: an observational study," *The Lancet*, vol. 365, pp. 1046-1053, 2005.
- [63] D. J. Eckert, A. S. Jordan, P. Merchia, and A. Malhotra, "Central sleep apnea: pathophysiology and treatment," *Chest Journal*, vol. 131, pp. 595-607, 2007.
- [64]F. Crummy, A. Piper, and M. Naughton, "Obesity and the lung: 2. Obesity and sleep-disordered breathing," *Thorax*, vol. 63, pp. 738-746, 2008.

- [65] H. K. Yaggi, J. Concato, W. N. Kernan, J. H. Lichtman, L. M. Brass, and V. Mohsenin, "Obstructive sleep apnea as a risk factor for stroke and death," *New England Journal of Medicine*, vol. 353, pp. 2034-2041, 2005.
- [66] S. Wilson and S. Argyropoulos, "Sleep in schizophrenia: time for closer attention," *The British Journal of Psychiatry*, vol. 200, pp. 273-274, 2012.
- [67] A. D. Krystal, "Psychiatric disorders and sleep," *Neurologic Clinics*, vol. 30, pp. 1389-1413, 2012.
- [68] S. Cohrs, "Sleep disturbances in patients with schizophrenia," *CNS drugs*, vol. 22, pp. 939-962, 2008.
- [69] A. Ilanković, A. Damjanović, V. Ilanković, B. Filipović, S. Janković, and N. Ilanković, "Polysomnographic sleep patterns in depressive, schizophrenic and healthy subjects," *Psychiatr Danub*, vol. 26, pp. 20-6, 2014.
- [70] K. L. Benson, E. V. Sullivan, K. O. Lim, J. Lauriello, V. P. Zarcone, and A. Pfefferbaum, "Slow wave sleep and computed tomographic measures of brain morphology in schizophrenia," *Psychiatry Research*, vol. 60, pp. 125-134, 1996.
- [71] I. Feinberg, P. Wender, R. Koresko, F. Gottlieb, and J. Piehuta, "Differential effects of chlorpromazine and phenobarbital on EEG sleep patterns," *Journal of Psychiatric Research*, vol. 7, pp. 101-109, 1969.
- [72] S. Giménez, S. Clos, S. Romero, E. Grasa, A. Morte, and M. J. Barbanoj, "Effects of olanzapine, risperidone and haloperidol on sleep after a single oral morning dose in healthy volunteers," *Psychopharmacology*, vol. 190, pp. 507-516, 2007.
- [73] N. Tsuno, A. Besset, and K. Ritchie, "Sleep and depression," *Journal of Clinical Psychiatry*, 2005.

- [74] D. J. Buysse, C. F. Reynolds, T. H. Monk, S. R. Berman, and D. J. Kupfer, "The Pittsburgh Sleep Quality Index: a new instrument for psychiatric practice and research," *Psychiatry Research*, vol. 28, pp. 193-213, 1989.
- [75] A. o. S. D. "Centers, Diagnostic classification of sleep and arousal disorders", *Raven*, 1979.
- [76] A. S. D. Association, "International Classification of Sleep Disorders: Diagnostic and Coding Manual. Revised Ed," ed: ASDA. Rochester, MA, 1997.
- [77] L. Colligan, Sleep Disorders: Marshall Cavendish Children's Books, 2008.
- [78]Z. Clemens, D. Fabó, and P. Halász, "Twenty-four hours retention of visuospatial memory correlates with the number of parietal sleep spindles," *Neuroscience Letters*, vol. 403, pp. 52-56, 2006.
- [79]Z. Clemens, D. Fabo, and P. Halasz, "Overnight verbal memory retention correlates with the number of sleep spindles," *Neuroscience*, vol. 132, pp. 529-535, 2005.
- [80] J. K. Matheson, R. Singh, and A. Packard, "Polysomnography and Sleep Disorders," in *The Clinical Neurophysiology Primer*, ed: Springer, pp. 393-445, 2007.
- [81]H. Schulz, "Rethinking sleep analysis: Comment on the AASM manual for the scoring of sleep and associated events," *Journal of Clinical Sleep Medicine: JCSM: Official Publication of the American Academy of Sleep Medicine,* vol. 4, p. 99, 2008.
- [82] A. Rodenbeck, R. Binder, P. Geisler, H. Danker-Hopfe, R. Lund, F. Raschke, *et al.*,
  "A review of sleep EEG patterns. Part I: a compilation of amended rules for their visual recognition according to Rechtschaffen and Kales," *Somnologie*, vol. 10, pp. 159-175, 2006.

- [83] H. Danker-hopfe, P. Anderer, J. Zeitlhofer, M. Boeck, H. Dorn, G. Gruber, *et al.*, "Interrater reliability for sleep scoring according to the Rechtschaffen & Kales and the new AASM standard," *Journal of Sleep Research*, vol. 18, pp. 74-84, 2009.
- [84] P. Anderer, G. Gruber, S. Parapatics, and G. Dorffner, "Automatic sleep classification according to Rechtschaffen and Kales," in *Engineering in Medicine and Biology Society*, 2007. EMBS 2007. 29th Annual International Conference of the IEEE, 2007, pp. 3994-3997.
- [85] J. Lawrence, et al., "Clinical guideline for the evaluation, management and longterm care of obstructive sleep apnea in adults," *Journal of clinical sleep medicine: JCSM: Official Publication of the American Academy of Sleep Medicine*, vol. 5, p. 263, 2009.
- [86] T. I. Morgenthaler, R. N. Aurora, T. Brown, R. Zak, C. Alessi, B. Boehlecke, *et al.*, "Practice parameters for the use of autotitrating continuous positive airway pressure devices for titrating pressures and treating adult patients with obstructive sleep apnea syndrome: an update for 2007: an American Academy of Sleep Medicine Report," *Sleep*, vol. 31, p. 141, 2008.
- [87] M. M. Grigg-Damberger, "The AASM scoring manual four years later," Journal of clinical sleep medicine: JCSM: Official Publication of the American Academy of Sleep Medicine, vol. 8, p. 323, 2012.
- [88] S. D. Davis, E. Eber, and A. C. Koumbourlis, *Diagnostic tests in pediatric pulmonology: Applications and Interpretation:* Springer, 2014.
- [89] T. L. Lee-Chiong, Sleep: a comprehensive handbook: Wiley Online Library, 2006.
- [90] S. Chokroverty, R. J. Thomas, and M. Bhatt, *Atlas of sleep medicine*: Elsevier Health Sciences, 2013.

- [91]C. A. Kushida, M. R. Littner, T. Morgenthaler, C. A. Alessi, D. Bailey, J. Coleman Jr, *et al.*, "Practice parameters for the indications for polysomnography and related procedures: an update for 2005," *Sleep*, vol. 28, pp. 499-521, 2005.
- [92] P. Panayiotopoulos, S. Benbadis, R. Beran, A. Berg, J. Engel Jr, and A. Galanopoulou, *Atlas of epilepsies* vol. 3: Springer Science & Business Media, 2010.
- [93] R. C. Basner, Case Studies in Polysomnography Interpretation: Cambridge University Press, 2012.
- [94] T. Roth, "Characteristics and determinants of normal sleep," *The Journal of Clinical Psychiatry*, vol. 65, pp. 8-11, 2003.
- [95] R. B. Berry, Fundamentals of sleep medicine: Elsevier Health Sciences, 2011.
- [96] E. Niedermeyer and F. L. da Silva, *Electroencephalography: basic principles*, *clinical applications, and related fields*: Lippincott Williams & Wilkins, 2005.
- [97] M. Oezgoeren, S. Kocaaslan, and A. Öniz, "Analysis of non-REM sleep staging with electroencephalography bispectral index," *Sleep and Biological Rhythms*, vol. 6, pp. 249-255, 2008.
- [98] W. O. Tatum, "Ellen R. Grass Lecture: Extraordinary EEG," *The Neurodiagnostic Journal*, vol. 54, pp. 3-21, 2014.
- [99] S. Murakami and Y. Okada, "Contributions of principal neocortical neurons to magnetoencephalography and electroencephalography signals," *The Journal of Physiology*, vol. 575, pp. 925-936, 2006.
- [100] D. K. Shah, M. T. Mackay, S. Lavery, S. Watson, A. S. Harvey, J. Zempel, *et al.*, "Accuracy of bedside electroencephalographic monitoring in comparison with simultaneous continuous conventional electroencephalography for seizure detection in term infants," *Pediatrics*, vol. 121, pp. 1146-1154, 2008.
- [101] S. Sanei and J. A. Chambers, *EEG signal processing*: John Wiley & Sons, 2013.

- [102] D. Foulkes, "Dreaming and REM sleep," *Journal of Sleep Research*, vol. 2, pp. 199-202, 1993.
- [103] J. d. R. Millán and J. Mouriño, "Asynchronous BCI and local neural classifiers: an overview of the adaptive brain interface project," *Neural Systems and Rehabilitation Engineering, IEEE Transactions on*, vol. 11, pp. 159-161, 2003.
- [104] G. H. Klem, H. O. Lüders, H. Jasper, and C. Elger, "The ten-twenty electrode system of the International Federation," *Electroencephalogr Clin Neurophysiol*, vol. 52, p. 3, 1999.
- [105] S. S. Agrawal ,V. M. Mane, "Bio-Medical Electronics with Oral Question and Answers", Nirali Prakashan, 2009.
- [106] S. Sadaghiani, R. Scheeringa, K. Lehongre, B. Morillon, A.-L. Giraud, and A. Kleinschmidt, "Intrinsic connectivity networks, alpha oscillations, and tonic alertness: a simultaneous electroencephalography/functional magnetic resonance imaging study," *The Journal of Neuroscience*, vol. 30, pp. 10243-10250, 2010.
- [107] R. B. Berry and M. H. Wagner, *Sleep Medicine Pearls: Expert Consult-Online*: Elsevier Health Sciences, 2014.
- [108] C. Pollak, M. J. Thorpy, and J. Yager, *The encyclopedia of sleep and sleep disorders*: InfoBase publishing, 2010.
- [109] S. Chokroverty, 100 Questions & Answers About Sleep and Sleep Disorders: Jones & Bartlett Publishers, 2007.
- [110] A. L. Loomis, E. N. Harvey, and G. Hobart, "Cerebral states during sleep, as studied by human brain potentials," *Journal of Experimental Psychology*, vol. 21, p. 127, 1937.
- [111] D. Hord, "Common mode rejection techniques in conjugate eye movement recording during sleep," *Psychophysiology*, vol. 12, pp. 354-355, 1975.

- [112] J. Hasan, "Differentiation of normal and disturbed sleep by automatic analysis," *Acta physiologica Scandinavica. Supplementum*, vol. 526, pp. 1-103, 1982.
- [113] T. C. Chieh, M. M. Mustafa, A. Hussain, S. F. Hendi, and B. Y. Majlis, "Development of vehicle driver drowsiness detection system using electrooculogram (EOG)," in *Computers, Communications, & Signal Processing with Special Track on Biomedical Engineering, 2005. CCSP 2005. 1st International Conference on, 2005, pp. 165-168.*
- [114] W. R. Miles, "Horizontal eye movements at the onset of sleep," *Psychological Review*, vol. 36, p. 122, 1929.
- [115] S. Chokroverty, R. P. Allen, A. S. Walters, and P. Montagna, *Sleep and movement disorders*: Oxford University Press, 2013.
- [116] J. Santamaria and K. H. Chiappa, "The EEG of drowsiness in normal adults," *Journal of Clinical Neurophysiology*, vol. 4, pp. 327-382, 1987.
- [117] D. Kirsch, *Sleep Medicine in Neurology*: John Wiley & Sons, 2013.
- [118] M. Atienza, J. L. Cantero, R. Stickgold, and J. Allan Hobson, "Eyelid movements measured by Nightcap predict slow eye movements during quiet wakefulness in humans," *Journal of Sleep Research*, vol. 13, pp. 25-29, 2004.
- [119] A. B. Usakli, S. Gurkan, F. Aloise, G. Vecchiato, and F. Babiloni, "On the use of electrooculogram for efficient human computer interfaces," *Computational Intelligence and Neuroscience*, vol. 2010, p. 1, 2010.
- [120] M. Fabbri, F. Provini, E. Magosso, A. Zaniboni, A. Bisulli, G. Plazzi, *et al.*,
   "Detection of sleep onset by analysis of slow eye movements: a preliminary study of MSLT recordings," *Sleep Medicine*, vol. 10, pp. 637-640, 2009.
- [121] A. H. Duc, P. Bays, and M. Husain, ".5-Eye movements as a probe of attention," *Progress in brain research*, vol. 171, pp. 403-411, 2008.

- [122] I. Gopal and G. Haddad, "Automatic detection of eye movements in REM sleep using the electrooculogram," *American Journal of Physiology-Regulatory*, *Integrative and Comparative Physiology*, vol. 241, pp. R217-R221, 1981.
- [123] Y. Tsuji, H. Satoh, N. Itoh, Y. Sekiguchi, and K. Nagasawa, "Automatic detection of rapid eye movements by discrete wavelet transform," *Psychiatry and Clinical Neurosciences*, vol. 54, pp. 276-277, 2000.
- [124] X. Tan, I. G. Campbell, and I. Feinberg, "A simple method for computer quantification of stage REM eye movement potentials," *Psychophysiology*, vol. 38, pp. 512-516, 2001.
- [125] R. Agarwal, T. Takeuchi, S. Laroche, and J. Gotman, "Detection of rapid-eye movements in sleep studies," *Biomedical Engineering, IEEE Transactions on*, vol. 52, pp. 1390-1396, 2005.
- [126] N. R. Burch, "Automatic analysis of the electroencephalogram: a review and classification of systems," *Electroencephalography and Clinical Neurophysiology*, vol. 11, pp. 827-834, 1959.
- [127] H. W. Agnew, J. C. Parker, W. B. Webb, and R. L. Williams, "Amplitude measurement of the sleep electroencephalogram," *Electroencephalography and Clinical Neurophysiology*, vol. 22, pp. 84-86, 1967.
- [128] J. D. Frost, "An automatic sleep analyzer," *Electroencephalography and Clinical Neurophysiology*, vol. 29, pp. 88-92, 1970.
- [129] P. Ktonas and A. Gosalia, "Spectral analysis vs. period-amplitude analysis of narrowband EEG activity: a comparison based on the sleep delta-frequency band," *Sleep*, vol. 4, pp. 193-206, 1980.
- [130] B. Ahmed and R. Tafreshi, "Advances in automatic sleep analysis," in *13th International Conference on Biomedical Engineering*, 2009, pp. 422-426.

- [131] L. Doroshenkov, V. Konyshev, and S. Selishchev, "Classification of human sleep stages based on EEG processing using hidden Markov models," *Biomedical Engineering*, vol. 41, pp. 25-28, 2007.
- [132] U. Malinowska, P. J. Durka, K. J. Blinowska, W. Szelenberger, and A. Wakarow, "Micro-and macrostructure of sleep EEG," *Engineering in Medicine and Biology Magazine, IEEE*, vol. 25, pp. 26-31, 2006.
- [133] R. G. Norman, I. Pal, C. Stewart, J. A. Walsleben, and D. M. Rapoport,
   "Interobserver agreement among sleep scorers from different centers in a large dataset," *Sleep*, vol. 23, pp. 901-908, 2000.
- [134] A. Piryatinska, G. Terdik, W. A. Woyczynski, K. A. Loparo, M. S. Scher, and A. Zlotnik, "Automated detection of neonate EEG sleep stages," *Computer Methods and Programs in Biomedicine*, vol. 95, pp. 31-46, 2009.
- [135] J. Caffarel, G. J. Gibson, J. P. Harrison, C. J. Griffiths, and M. J. Drinnan, "Comparison of manual sleep staging with automated neural network-based analysis in clinical practice," *Medical and Biological Engineering and Computing*, vol. 44, pp. 105-110, 2006.
- [136] W. Martin, L. Johnson, S. Viglione, P. Naitoh, R. Joseph, and J. Moses, "Pattern recognition of EEG-EOG as a technique for all-night sleep stage scoring," *Electroencephalography and Clinical Neurophysiology*, vol. 32, pp. 417-427, 1972.
- [137] J. Virkkala, J. Hasan, A. Varri, S.-L. Himanen, and K. Muller, "Automatic sleep stage classification using two-channel electro-oculography," *Journal of Neuroscience Methods*, vol. 166, pp. 109-115, 2007.
- [138] E. Stanus, B. Lacroix, M. Kerkhofs, and J. Mendlewicz, "Automated sleep scoring: a comparative reliability study of two algorithms," *Electroencephalography and Clinical Neurophysiology*, vol. 66, pp. 448-456, 1987.

- [139] E. Huupponen, S.-L. Himanen, J. Hasan, and A. Värri, "Sleep depth oscillations: an aspect to consider in automatic sleep analysis," *Journal of Medical Systems*, vol. 27, pp. 337-345, 2003.
- [140] A. Saastamoinen, E. Huupponen, A. Värri, J. Hasan, and S.-L. Himanen,
   "Computer program for automated sleep depth estimation," *Computer Methods and Programs in Biomedicine*, vol. 82, pp. 58-66, 2006.
- [141] E. Estrada, H. Nazeran, P. Nava, K. Behbehani, J. Burk, and E. Lucas, "EEG feature extraction for classification of sleep stages," in *Engineering in Medicine and Biology Society, 2004. IEMBS'04. 26th Annual International Conference of the IEEE*, 2004, pp. 196-199.
- [142] J. R. Smith and I. Karacan, "EEG sleep stage scoring by an automatic hybrid system," *Electroencephalography and Clinical Neurophysiology*, vol. 31, pp. 231-237, 1971.
- [143] L. Fraiwan, K. Lweesy, N. Khasawneh, M. Fraiwan, H. Wenz, and H. Dickhaus,
   "Classification of sleep stages using multi-wavelet time frequency entropy and LDA," *Methods of information in Medicine*, vol. 49, p. 230, 2010.
- [144] A. Takajyo, M. Katayama, K. Inoue, K. Kumamaru, and S. Matsuoka, "Timefrequency analysis of human sleep EEG," in *SICE-ICASE*, 2006. International Joint Conference, pp. 3303-3307, 2006.
- [145] C.-F. Chao, J.-A. Jiang, M.-J. Chiu, and R.-G. Lee, "Wavelet-based processing and adaptive fuzzy clustering for automated long-term polysomnography analysis," in *Acoustics, Speech and Signal Processing, 2006. ICASSP 2006 Proceedings. 2006 IEEE International Conference on*, pp. II-II,2006.

- [146] H. Ocak, "Automatic detection of epileptic seizures in EEG using discrete wavelet transform and approximate entropy," *Expert Systems with Applications*, vol. 36, pp. 2027-2036, 2009.
- [147] S. V. Schönwald, L. Emerson, R. Rossatto, M. L. Chaves, and G. J. Gerhardt,
   "Benchmarking matching pursuit to find sleep spindles," *Journal of Neuroscience Methods*, vol. 156, pp. 314-321, 2006.
- [148] H. Park, K. Park, and D.-U. Jeong, "Hybrid neural-network and rule-based expert system for automatic sleep stage scoring," in *Engineering in Medicine and Biology Society, 2000. Proceedings of the 22nd Annual International Conference of the IEEE*, 2000, pp. 1316-1319.
- [149] M. Schwaibold, T. Penzel, J. Schochlin, and A. Bolz, "Combination of AI components for biosignal processing application to sleep stage recognition," in *Engineering in Medicine and Biology Society, 2001. Proceedings of the 23rd Annual International Conference of the IEEE*, 2001, pp. 1692-1694.
- [150] P. Anderer, G. Gruber, S. Parapatics, M. Woertz, T. Miazhynskaia, G. Klösch, et al., "An E-health solution for automatic sleep classification according to Rechtschaffen and Kales: validation study of the Somnolyzer 24× 7 utilizing the Siesta database," *Neuropsychobiology*, vol. 51, pp. 115-133, 2005.
- [151] R. Agarwal and J. Gotman, "Computer-assisted sleep staging," *Biomedical Engineering, IEEE Transactions on*, vol. 48, pp. 1412-1423, 2001.
- [152] T. Schlüter and S. Conrad, "An approach for automatic sleep stage scoring and apnea-hypopnea detection," *Frontiers of Computer Science*, vol. 6, pp. 230-241, 2012.

- [153] K. Šušmáková and A. Krakovská, "Discrimination ability of individual measures used in sleep stages classification," *Artificial Intelligence in Medicine*, vol. 44, pp. 261-277, 2008.
- [154] I. Zhovna and I. D. Shallom, "Automatic detection and classification of sleep stages by multichannel EEG signal modeling," in *Engineering in Medicine and Biology Society*, 2008. EMBS 2008. 30th Annual International Conference of the IEEE, 2008, pp. 2665-2668.
- [155] S. D. Pittman, M. M. MacDonald, R. B. Fogel, A. Malhotra, K. Todros, B. Levy, et al., "Assessment of automated scoring of polysomnographic recordings in a population with suspected sleep-disordered breathing," *Sleep-New York then Westchester-*, vol. 27, pp. 1394-1404, 2004.
- [156] L. H. Larsen, K. E. Moe, M. V. Vitiello, and P. N. Prinz, "A note on the nightto-night stability of stages 3+ 4 sleep in healthy older adults: a comparison of visual and spectral evaluations of stages 3+ 4 sleep," *Sleep*, vol. 18, pp. 7-10, 1995.
- [157] P. Van Hese, W. Philips, J. De Koninck, R. Van de Walle, and I. Lemahieu, "Automatic detection of sleep stages using the EEG," in *Engineering in Medicine* and Biology Society, 2001. Proceedings of the 23rd Annual International Conference of the IEEE, 2001, pp. 1944-1947.
- [158] Z. Zhang, C. Guan, T. E. Chan, J. Yu, A. K. Ng, H. Zhang, et al., "Automatic sleep onset detection using single EEG sensor," in *Engineering in Medicine and Biology Society (EMBC), 2014 36th Annual International Conference of the IEEE*, 2014, pp. 2265-2268.
- [159] R. S. Huang, C. J. Kuo, L.-L. Tsai, and O. T. Chen, "EEG pattern recognitionarousal states detection and classification," in *Neural Networks*, 1996., IEEE *International Conference on*, 1996, pp. 641-646.

- [160] D. Novak, Y. Al-Ani, and L. Lhotska, "Electroencephalogram processing using hidden markov models," *EUROSIM'04*, 2004.
- [161] A. B. Rossow, E. O. T. Salles, and K. F. Côco, "Automatic sleep staging using a single-channel EEG modeling by Kalman filter and HMM," in *Biosignals and Biorobotics Conference (BRC)*, 2011 ISSNIP, 2011, pp. 1-6.
- [162] N. Sukhorukova, A. Stranieri, B. Ofoghi, P. Vamplew, M. Saleem, L. Ma, et al., "Automatic sleep stage identification: difficulties and possible solutions," in Proceedings of the Fourth Australasian Workshop on Health Informatics and Knowledge Management-Volume 108, 2010, pp. 39-44.
- [163] T. Hori, M. Hayashi, and T. Morikawa, "Topographical EEG changes and the hypnagogic experience," 1994.
- [164] S.-L. Himanen, A new visual adaptive scoring system for sleep recordings. Development and application to the multiple sleep latency test: Tampere University Press, 2000.
- [165] J. C. Principe, S. K. Gala, and T. G. Chang, "Sleep staging automaton based on the theory of evidence," *Biomedical Engineering, IEEE Transactions on*, vol. 36, pp. 503-509, 1989.
- [166] N. McGrogan, E. Braithwaite, and L. Tarassenko, "Biosleep: a comprehensive sleep analysis system," in *Engineering in Medicine and Biology Society*, 2001. *Proceedings of the 23rd Annual International Conference of the IEEE*, 2001, pp. 1608-1611.
- [167] A. Flexer, G. Gruber, and G. Dorffner, "Improvements on continuous unsupervised sleep staging," in *Neural Networks for Signal Processing*, 2002. *Proceedings of the 2002 12th IEEE Workshop on*, 2002, pp. 687-695.

- [168] M. Hanaoka, M. Kobayashi, and H. Yamazaki, "Automatic sleep stage scoring based on waveform recognition method and decision-tree learning," *Systems and Computers in Japan*, vol. 33, pp. 1-13, 2002.
- [169] L. Huang, Q. Sun, and J. Cheng, "Novel method of fast automated discrimination of sleep stages," in *Engineering in Medicine and Biology Society*, 2003. Proceedings of the 25th Annual International Conference of the IEEE, 2003, pp. 2273-2276.
- [170] E. Huupponen, S.-L. Himanen, J. Hasan, and A. Värri, "Sleep depth oscillations: an aspect to consider in automatic sleep analysis," *Journal of Medical Systems*, vol. 27, pp. 337-345, 2003.
- [171] S. Gudmundsson, T. P. Runarsson, and S. Sigurdsson, "Automatic sleep staging using support vector machines with posterior probability estimates," in *Computational Intelligence for Modelling, Control and Automation, 2005 and International Conference on Intelligent Agents, Web Technologies and Internet Commerce, International Conference on, 2005, pp. 366-372.*
- [172] I. Song, Y. Ji, B. Cho, J. Ku, Y. Chee, J. Lee, *et al.*, "Multifractal analysis of sleep EEG dynamics in humans," in *Neural Engineering*, 2007. CNE'07. 3rd International IEEE/EMBS Conference on, 2007, pp. 546-549.
- [173] S. Güneş, K. Polat, and Ş. Yosunkaya, "Efficient sleep stage recognition system based on EEG signal using k-means clustering based feature weighting," *Expert Systems with Applications*, vol. 37, pp. 7922-7928, 2010.
- [174] J. Hedner, D. P. White, A. Malhotra, S. Herscovici, S. D. Pittman, D. Zou, et al., "Sleep staging based on autonomic signals: a multi-center validation study," Journal of clinical sleep medicine: JCSM: official publication of the American Academy of Sleep Medicine, vol. 7, p. 301, 2011.

- [175] J. Dong, D. Liu, C. Zhang, J. Ma, G. Wang, D. Guo, *et al.*, "Automated sleep staging technique based on the empirical mode decomposition algorithm: a preliminary study," *Advances in Adaptive Data Analysis*, vol. 2, pp. 267-276, 2010.
- [176] S.-F. Liang, C.-E. Kuo, Y.-H. Hu, Y.-H. Pan, and Y.-H. Wang, "Automatic stage scoring of single-channel sleep EEG by using multiscale entropy and autoregressive models," *Instrumentation and Measurement, IEEE Transactions on*, vol. 61, pp. 1649-1657, 2012.
- [177] E. Tagliazucchi, F. von Wegner, A. Morzelewski, S. Borisov, K. Jahnke, and
   H. Laufs, "Automatic sleep staging using fMRI functional connectivity data," *Neuroimage*, vol. 63, pp. 63-72, 2012.
- [178] D. Álvarez-Estévez, J. M. Fernández-Pastoriza, E. Hernández-Pereira, and V. Moret-Bonillo, "A method for the automatic analysis of the sleep macrostructure in continuum," *Expert Systems with Applications*, vol. 40, pp. 1796-1803, 2013.
- [179] A. G. Correa, L. Orosco, and E. Laciar, "Automatic detection of drowsiness in EEG records based on multimodal analysis," *Medical Engineering & Physics*, vol. 36, pp. 244-249, 2014.
- [180] D. Popovic, M. Khoo, and P. Westbrook, "Automatic scoring of sleep stages and cortical arousals using two electrodes on the forehead: validation in healthy adults," *Journal of Sleep Research*, vol. 23, pp. 211-221, 2014.
- [181] S. Devuyst," The Dreams databases" 2005. [Online]. Available: http://www.tcts.fpms.ac.be/~devuyst/. [Accessed: Oct. 12, 2015].
- [182] A. Flexer, G. Gruber, and G. Dorffner, "A reliable probabilistic sleep stager based on a single EEG signal," *Artificial Intelligence in Medicine*, vol. 33, pp. 199-207, 2005.

- [183] M. Ghil, "The SSA-MTM Toolkit: Applications to analysis and prediction of time series," in *Optical Science, Engineering and Instrumentation*'97, pp. 216-230, 1997.
- [184] M. Alipoor, M. Pooyan, and A. A. Suratgar, "Classification of EEG signals in four groups, including healthy subjects with open/closed eyes and epilepsy subjects with/without seizure by PSD estimate (using the multitaper method) and ANN," in *Health Informatics and Bioinformatics (HIBIT), 2010 5th International Symposium* on, 2010, pp. 98-103.
- [185] S. L. Marple Jr, "Digital spectral analysis with applications," *Englewood Cliffs*, *NJ*, *Prentice-Hall*, *Inc.*, *1987*, vol. 1, *512 p*. 1987.
- [186] E. Malaekah and D. Cvetkovic, "Automatic sleep stage detection using consecutive and non-consecutive approach for elderly and young healthy subject," in *Biosignals and Biorobotics Conference (2014): Biosignals and Robotics for Better and Safer Living (BRC), 5th ISSNIP-IEEE*, 2014, pp. 1-6.
- [187] H. Akaike, "A Bayesian extension of the minimum AIC procedure of autoregressive model fitting," *Biometrika*, vol. 66, pp. 237-242, 1979.
- [188] S. Uchida, M. Matsuura, S. Ogata, T. Yamamoto, and N. Aikawa, "Computerization of Fujimori's method of waveform recognition A review and methodological considerations for its application to all-night sleep EEG," *Journal of Neuroscience Methods*, vol. 64, pp. 1-12, 1996.
- [189] A. Krakovská and K. Mezeiová, "Automatic sleep scoring: A search for an optimal combination of measures," *Artificial Intelligence in Medicine*, vol. 53, pp. 25-33, 2011.
- [190] H. Kuwahara, H. Higashi, Y. Mizuki, S. Matsunari, M. Tanaka, and K. Inanaga, "Automatic real-time analysis of human sleep stages by an interval histogram

method," *Electroencephalography and Clinical Neurophysiology*, vol. 70, pp. 220-229, 1988.

- [191] A. G. Correa, E. Laciar, H. Patiño, and M. Valentinuzzi, "Artifact removal from EEG signals using adaptive filters in cascade," in *Journal of Physics: Conference Series*, 2007, p. 012081.
- [192] M. Mustafa, M. RTaid, Z. H. Murat, and N. Sulaiman, "Comparison between KNN and ANN classification in brain balancing application via spectrogram image," *JCSCM*, vol. 2, pp. 17-22, 2012.
- [193] S. Devuyst, T. Dutoit, P. Stenuit, and M. Kerkhofs, "Automatic sleep spindles detection—overview and development of a standard proposal assessment method," in *Engineering in Medicine and Biology Society, EMBC, 2011 Annual International Conference of the IEEE*, 2011, pp. 1713-1716.
- [194] J. Cohen, "A coefficient of agreement for nominal scales," *Educational and Psychological Measurement*, vol. 20, pp. 37-46, 1960.
- [195] T. Penzel and R. Conradt, "Computer based sleep recording and analysis," *Sleep Medicine Reviews*, vol. 4, pp. 131-148, 2000.
- [196] A. Flexer, P. Sykacek, I. Rezek, and G. Dorffner, "Using hidden Markov models to build an automatic, continuous and probabilistic sleep stager," in *Neural Networks, 2000. IJCNN 2000, Proceedings of the IEEE-INNS-ENNS International Joint Conference on*, 2000, pp. 627-631.
- [197] C. Berthomier, X. Drouot, M. Herman-Stoïca, P. Berthomier, J. Prado, D. Bokar-Thire, *et al.*, "Automatic analysis of single-channel sleep EEG: validation in healthy individuals," *Sleep*, vol. 30, p. 1587, 2007.
- [198] D. Aeschbach and A. A. Borbély, "All-night dynamics of the human sleep EEG," *Journal of Sleep Research*, vol. 2, pp. 70-81, 1993.

- [199] D. Popovic, I. Ayappa, P. Hauri, D. Levendowski, V. Velimirovic, O. Burschtin, *et al.*, "Accuracy of automated sleep staging using signals from a single forehead site," *Sleep*, vol. 31, p. 1007, 2008.
- [200] J. R. Shambroom, S. E. Fabregas, and J. Johnstone, "Validation of an automated wireless system to monitor sleep in healthy adults," *Journal of Sleep Research*, vol. 21, pp. 221-230, 2012.
- [201] E. Malaekah and D. Cvetkovic, "Automatic detection of the wake and stage 1 sleep stages using the EEG sub-epoch approach," in *Engineering in Medicine and Biology Society (EMBC), 2013 35th Annual International Conference of the IEEE*, 2013, pp. 6401-6404.
- [202] S. Chouinard, J. Poulin, E. Stip, and R. Godbout, "Sleep in untreated patients with schizophrenia: a meta-analysis," *Schizophrenia Bulletin*, vol. 30, p. 957, 2004.
- [203] M. S. Keshavan, C. F. Reynolds, J. M. Miewald, D. M. Montrose, J. A. Sweeney, R. C. Vasko, *et al.*, "Delta sleep deficits in schizophrenia: evidence from automated analyses of sleep data," *Archives of General Psychiatry*, vol. 55, pp. 443-448, 1998.
- [204] S. Nishino, E. Mignot, K. L. Benson, and V. P. Zarcone Jr, "Cerebrospinal fluid prostaglandins and corticotropin releasing factor in schizophrenics and controls: relationship to sleep architecture," *Psychiatry Research*, vol. 78, pp. 141-150, 1998.
- [205] Y. Doi, M. Minowa, M. Okawa, and M. Uchiyama, "Prevalence of sleep disturbance and hypnotic medication use in relation to sociodemographic factors in the general Japanese adult population," *Journal of Epidemiology*, vol. 10, pp. 79-86, 2000.
